

A Service of



Leibniz-Informationszentrum Wirtschaft Leibniz Information Centre for Economics

Oswald-Egg, Maria Esther; Siegenthaler, Michael

Working Paper Train drain? Access to skilled foreign workers and firms' provision of training

KOF Working Papers, No. 495

Provided in Cooperation with: KOF Swiss Economic Institute, ETH Zurich

Suggested Citation: Oswald-Egg, Maria Esther; Siegenthaler, Michael (2021) : Train drain? Access to skilled foreign workers and firms' provision of training, KOF Working Papers, No. 495, ETH Zurich, KOF Swiss Economic Institute, Zurich, https://doi.org/10.3929/ethz-b-000501254

This Version is available at: https://hdl.handle.net/10419/251035

Standard-Nutzungsbedingungen:

Die Dokumente auf EconStor dürfen zu eigenen wissenschaftlichen Zwecken und zum Privatgebrauch gespeichert und kopiert werden.

Sie dürfen die Dokumente nicht für öffentliche oder kommerzielle Zwecke vervielfältigen, öffentlich ausstellen, öffentlich zugänglich machen, vertreiben oder anderweitig nutzen.

Sofern die Verfasser die Dokumente unter Open-Content-Lizenzen (insbesondere CC-Lizenzen) zur Verfügung gestellt haben sollten, gelten abweichend von diesen Nutzungsbedingungen die in der dort genannten Lizenz gewährten Nutzungsrechte.

Terms of use:

Documents in EconStor may be saved and copied for your personal and scholarly purposes.

You are not to copy documents for public or commercial purposes, to exhibit the documents publicly, to make them publicly available on the internet, or to distribute or otherwise use the documents in public.

If the documents have been made available under an Open Content Licence (especially Creative Commons Licences), you may exercise further usage rights as specified in the indicated licence.



WWW.ECONSTOR.EU

EHzürich

Research Collection



Working Paper

Train drain? Access to skilled foreign workers and firms' provision of training

Author(s): Oswald-Egg, Maria Esther; Siegenthaler, Michael

Publication Date: 2021-08

Permanent Link: https://doi.org/10.3929/ethz-b-000501254 →

Rights / License: In Copyright - Non-Commercial Use Permitted →

This page was generated automatically upon download from the <u>ETH Zurich Research Collection</u>. For more information please consult the <u>Terms of use</u>.

ETH zürich

KOF Swiss Economic Institute

Train drain? Access to skilled foreign workers and firms' provision of training

Maria Esther Oswald-Egg and Michael Siegenthaler

KOF Working Papers, No. 495, August 2021

KOF

ETH Zurich KOF Swiss Economic Institute LEE G 116 Leonhardstrasse 21 8092 Zurich, Switzerland

Phone +41 44 632 42 39 Fax +41 44 632 12 18 www.kof.ethz.ch kof@kof.ethz.ch

Train drain? Access to skilled foreign workers and firms' provision of training

Maria Esther Oswald-Egg Michael Siegenthaler*

August 10, 2021

Abstract

Does better access to skilled workers reduce firms' willingness to provide general skills training to unskilled workers? We analyze how the gradual opening of the Swiss labor market to workers from the European Union affected the number of apprenticeship positions that firms provide. We exploit that the availability of skilled workers increased more in firms close to the border because they gained unrestricted access to cross-border workers from neighboring countries. Our Difference-in-Differences estimates suggest that firm-provided training and access to skilled workers are not necessarily substitutes: opening the borders did not have a statistically significant effect on apprenticeship provision. We show theoretically and empirically that the small impact was the consequence of two opposing effects: the greater availability of skilled workers reduced firms' incentive to train because the cost of hiring external labor fell. Positive impacts on firm growth worked in the opposite direction.

Keywords: labor demand; skilled immigration; firm-provided training; apprenticeships; vocational education and training; free movement of workers; cross-border workers; recruitment; immigration policy; labor mobility; hiring costs

JEL: J24, J63, M53

^{*}ETH Zurich, KOF Swiss Economic Institute, Leonhardstrasse 21, CH–8092 Zurich, Switzerland. Corresponding author: Michael Siegenthaler, siegenthaler@kof.ethz.ch. We are grateful to Stefan Wolter for providing the cost-benefit data and his colleagues at SFIVET, who helped match the data with the firms' geocodes. We thank Uschi Backes-Gellner, Andreas Beerli, Thomas Bolli, Daniel Kopp, Harald Pfeifer, Ursula Renold, Jan-Egbert Sturm, Stefan Wolter, the participants of the KOF Brown Bag Seminar, and the members of the Swiss Leading House on Economics of Education for helpful comments on earlier versions of the paper.

1 Introduction

Does firms' willingness to provide general skills training depend on the supply of skilled workers in the local labor market? A greater local supply makes it easier and cheaper for firms to find external workers with the right skills, which could reduce their incentive to invest in the skills of their internal workers. Indeed, both the theoretical and empirical literature on firm-sponsored general training highlights a potential trade-off between the supply of skilled workers and firms' willingness to train.¹ If present, such a trade-off would have important implications for policy makers. A case in point is immigration policy. If training and the local supply of skilled workers are substitutes, removing migration restrictions would shift the costs of acquiring labor market skills from firms to native workers and firms and governments in immigrants' source countries. Such shifts could reinforce a possible brain drain associated with free mobility of labor and may lead to opposition among residents against open borders.²

This paper provides, to the best of our knowledge, the first quasi-experimental evidence whether the local supply of skilled workers affects firms' provision of general training. We exploit an exogenous increase in the availability of skilled workers that resulted from the implementation of the Agreement on the Free Movement of unrestricted access to the Swiss labor market starting from 1999 onward. As shown by Beerli et al. (2021), firms near the border incidentally experienced a larger shock because the free movement policy abolished the pre-existing restrictions on hiring and employing cross-border workers from Switzerland's neighboring countries. Cross-border workers work almost exclusively in regions within 30 minutes' commuting time of the border. Thus, firms close to the border hired substantially more EU workers after the border opening than firms farther away from the border. Following Beerli et al. (2021), we thus study the policy's effects using a transparent

¹In the training model of Stevens (1994), firms train skilled workers to save on hiring costs for skilled workers in the external labor market. Similarly, in the search and matching models of Shintoyo (2008, 2010), the proportion of skilled workers in the unemployment pool is the central (endogenous) determinant of firm training. If skilled labor is abundant, firms prefer external recruitment to training. Existing empirical papers tend to support the notion of a trade-off between external recruitment and training. For instance, Blatter et al. (2012, 2016) document a positive correlation between the costs of recruiting workers from the external labour market and firms' provision of general skills training.

²Denmark, for example, defends its strict immigration policies with the argument that Danish firms should invest more in the training of native workers to cover their hiring needs (cf. Hermann, 2019).

Differences-in-Differences design that compares changes in firms' provision of general skills training near the border—the treatment group—with changes in firms' training farther away. We measure general skills training by looking at firms' provision of apprenticeships for young adults.³

In the cross-section, apprenticeship provision depends strikingly on the distance to the nearest border crossing. Using data from the Swiss Business Censuses 1995– 2008 that provide the employment and geo-coordinates for the universe of establishments in Switzerland, we find that firms located near the border are substantially less likely to train apprentices. In contrast, these firms are more likely to employ cross-border workers—and foreign workers in total—than firms farther away. However, these cross-sectional relationships do not appear to be causal. Our DiD estimates suggest that the free movement policy had a substantial and highly statistically significant positive impact on the employment of foreign workers in firms close to the border in general and among firms that train apprentices in particular. However, analogous effect estimates on the number of apprenticeship positions are statistically indistinguishable from zero, even among highly treated establishments within 15 minutes of travel time to the border. This conclusion holds in various robustness checks, including a matching approach that only compares establishments with similar pre-reform characteristics.

We interpret these results through the lens of a basic dynamic labor-demand model of training originally proposed by Stevens (1994). We extend this model to allow for skilled immigration. Central to firms' dynamic decisions are the assumptions that firms face exogenous separations from skilled workers and costs for replacing them. Firms thus train unskilled workers today to save on hiring costs for skilled workers tomorrow, in line with empirical evidence that the supply of training increases with hiring costs (e.g., Blatter et al., 2016). We also assume that firms produce with a constant elasticity of substitution (CES) production function with two labor inputs: skilled immigrant workers and native workers. The CES framework allows the two labor inputs to be imperfect substitutes, a standard assumption in the economics of immigration (e.g., Ottaviano and Peri, 2012). In this

³Apprenticeships are an upper-secondary level education program combining formal schooling with work-based practical learning in firms (OECD, 2004). Apprenticeships in Switzerland are largely composed of general (i.e., transferable) skills (Mueller and Schweri, 2015). Therefore, many studies used the firms' employment of apprentices to study firms' willingness to provide general training, including the influential study by Acemoglu and Pischke (1998). See Wolter and Ryan (2011) for an overview of the economic literature on apprenticeship training.

framework, a greater availability of skilled immigrant workers does not necessarily reduce firms' provision of training because there are two countervailing effects. On the one hand, better access to suitable workers lowers the future savings in terms of hiring costs associated with training (*cost effect*). On the other hand, if native and immigrant workers are imperfect substitutes, the additional hiring of skilled foreign workers increases firms' demand for all types of workers through a reduction in wage costs (the *scale effect* familiar from standard labor demand theory).⁴

We then provide empirical evidence that the two mechanisms are at play, and counteract each other, in our setting. To this end, we exploit rich firm-level surveys in 2000, 2004, and 2009 on the costs, benefits, and motives of apprenticeship provision in Switzerland (Schweri et al., 2003; Muehlemann et al., 2007; Strupler and Wolter, 2012). Consistent with the cost effect, the free movement policy reduced firms' willingness to train to attract skilled workers because it is hard to find them elsewhere. Firms also became significantly less likely to train to save hiring costs for external workers. Consistent with the scale effect, the free movement policy had a quantitatively meaningful effect on establishment size in manufacturing and a sizeable but marginally insignificant effect in the private service sector. We find evidence for a displacement of apprenticeship positions in the construction sector, precisely the sector where we do not observe a positive effect on firm growth.

Our study contributes to the literature on immigration and firm-provided training in four important ways. First, we provide causal evidence on one particular factor that shapes firms' decisions to train unskilled workers instead of hiring skilled external ones. Few empirical studies have examined the determinants of these "make or buy"-decisions of firms.⁵ Our conceptual framework highlights that firms' costbenefit considerations and production complementarities shape the effects of the supply of skilled workers on firms' provision of training. These findings provide the mirror image to the empirical evidence that companies are discouraged from

⁴An alternative transmission channel is that skilled foreign immigrants increase firms' size by increasing innovation output and productivity, and hence total factor productivity. Indeed, Beerli et al. (2021) provide evidence that this alternative channel may be at work in our empirical setting, too.

⁵Examples of observational studies that analyze the determinants of "make or buy" decisions are Bellmann et al. (2014) and Blatter et al. (2016). However, overview articles on why firms train such as Leuven (2005) and Bassanini et al. (2007) do not touch directly upon these decisions. Training models typically assume that firms are homogeneous in terms of cost structure as well as training and recruitment cost function. Hence, either all firms train or no firm does. Once we relax these restrictive assumptions, firms' optimal recruitment strategies may differ and consist of both hiring externally and supplying some internal training (Wolter and Ryan, 2011).

training because other companies might poach trained workers (e.g., Winkelmann, 1996; Cappelli, 2004; Mohrenweiser and Zwick, 2009; Muehlemann and Wolter, 2011; Mohrenweiser et al., 2019). Conceptually, these papers analyze changes in the number of firms competing for fixed number of workers, while we study changes in the number of skilled workers for an initially fixed number of firms.

Second, our study provides the first quasi-experimental evidence on how firms' willingness to train unskilled workers depends on immigration.⁶ The existing empirical studies on this relationship are observational and reach conflicting conclusions. The reported estimates suggest that more immigration has no (Baker and Wooden, 1992), a positive (Campo et al., 2018), or a negative effect (Mountford and Wadsworth, 2019; Aepli and Kuhn, 2021) on training provision.⁷ Our main contribution relative to these papers is that we exploit quasi-experimental variation in the availability of skilled workers to identify the causal impact of better access to skilled foreign workers on training. We also contribute to this literature by providing direct evidence on two potential mechanisms underlying the effects of better access to skilled foreign workers on training provision. Our simple theoretical framework complements the model proposed by Mountford and Wadsworth (2019) by focusing on the within-firm effects of immigration on training. The model of Mountford and Wadsworth (2019) instead focuses on the between-firm (reallocation) effects.

Importantly, our main finding that firm-provided training and access to skilled workers are not necessarily substitutes stands in contrast to the results of a concurrent study by Aepli and Kuhn (2021). Using Swiss business census data, they argue that the employment of cross-border workers significantly replaces firms' provision of apprenticeships. The difference in results relative to ours is due to the differences in research design. Aepli and Kuhn (2021) estimate the effect of cross-border workers on training by instrumenting firms' employment of cross-border workers with an establishment's distance from the border. The exclusion restriction of this instrumental variable strategy states that distance from the border affects firms'

⁶There is a larger literature that analyzes the effects of immigration on the educational choices of residents. Two recent examples are Bächli and Tsankova (2020) and Brunello et al. (2020).

⁷Using Australian data, Baker and Wooden (1992) report a negative correlation between immigration and in-house training. This association, however, disappears once the authors control for industry fixed effects. Mountford and Wadsworth (2019) find a negative association between immigration and the share of UK-born workers that receive on-the-job training in sector times occupation cells. Finally, Campo et al. (2018) find a positive impact of skilled immigration on onand off-the-job training acquired by workers in the UK. Their identification strategy relies on a standard spatial shift-share instrumental variable approach following Card (2001). See Jaeger et al. (2018) for a critical assessment of this strategy.

provision of training *only* through the greater supply of cross-border workers near the border. Our DiD approach, which exploits quasi-experimental variation in the availability of cross-border workers, suggests that part of the differences in training behavior between firms close and farther away from the border are unrelated to the supply of cross-border workers.

Third, by focusing on firms' provision of training, our study also adds a novel angle to recent studies looking at the effects of immigration on wages and employment (Mansour, 2010; Dustmann et al., 2017) and on firms (Beerli et al., 2021) by exploiting changes in commuting policies. An attractive feature of this approach is that the increase in immigrant workers and its unequal regional impact are a direct consequence of the exogenous change in immigration policy. This approach contrasts the traditional shift-share instrumental variable approach to isolate supply-driven variation in regional immigration flows, which hinges on the assumption that historical immigrant concentrations across regions are uncorrelated with the current regional distribution of labor-demand shocks. This assumption is not always warranted (Jaeger et al., 2018). Another difference is that possible consumption-side effects of immigration (e.g., on demand for local housing) are muted in the case of commuting policies because cross-border workers do not relocate to the country they work. While this absence of reallocation likely increases the scope for displacement effects in the labor market, it also makes it easier to isolate the cost and scale effects highlighted by our demand-side framework.

Fourth, by linking our empirical results to the introduction of the free movement policy, our study has important implications for policy makers on the potential societal costs of training opportunities for young adults in the receiving country particularly given growing opposition to the free movement of workers in several European countries. In Switzerland, with widespread fears that immigration may harm the economic opportunities of citizens, the majority of Swiss citizens voted for a 2014 referendum—the "stop mass migration" initiative—which was intended to restrict the free movement of persons. One argument in the pre-voting debate was that open borders incentivize firms to hire cheap labor from abroad (Blocher, 2011), thereby undermining firms' willingness to invest in the training of unskilled citizens and reinforcing the brain drain caused by skilled migration.

This study is organized as follows. Section 2 presents a theoretical framework that guides our empirical analyses. Section 3 provides the institutional background on the reform process of the free movement policy and the apprenticeship system in Switzerland. Section 4 describes the data in the empirical part and the research design. Section 5 presents the empirical results and important robustness checks. Section 6 discusses our main findings and concludes.

2 Theoretical framework

Our theoretical framework is based on a dynamic labor-demand model with adjustment costs augmented with firms' investment in training (Stevens, 1994). In the model, firms can recruit skilled workers either by hiring them externally tomorrow at certain hiring costs or by training unskilled workers internally today at certain training costs. The model formalizes the investment approach to firmprovided training (Oatey, 1970; Lindley, 1975; Merrilees, 1983). We extend the original model of Stevens (1994) to study the effects of a greater availability of skilled immigrant workers. We omit the discount rate because it is not affected by immigration, and we focus on only two periods $t \in \{0, 1\}$. These simplifications make the exposition easier.

The model has three types of workers: unskilled native workers T_t , skilled native workers N_t , and skilled immigrant workers I_t . Firms hire skilled workers $L_t = N_t + I_t$ in both periods, and they can choose to hire unskilled workers (T_t) in the first period to train them and thus have skilled workers for the second period. There are constant and exogenous voluntary quit rates for trained unskilled workers (γ) and skilled workers (δ) . Firms' skilled employment in the second period, L_1 , therefore consists of the skilled workers remaining from the first period, the remaining trained unskilled workers from the first period T_0 (newly trained skilled workers), and skilled workers (X_1) newly hired from the external labor market:

$$L_1 = N_1 + I_1 \tag{1a}$$

with $N_1 = (1 - \delta)N_0 + (1 - \gamma)T_0 + X_1^N$ (1b)

and
$$I_1 = (1 - \delta)I_0 + X_1^I$$
 (1c)

Following Stevens (1994), we assume that unskilled native workers do not produce output in the training period. Thus, the firm's revenue function depends only on skilled labor $(R_t[L_t] \text{ with } R' > 0, R'' \leq 0)$. The price is set to unity. The revenue function hence equals the production function. In line with the previous literature, skilled native and skilled immigrant workers may be imperfect substitutes.⁸ Thus, we assume that firms produce with a constant elasticity of substitution (CES) production function with inputs I_t and N_t . The degree of imperfect substitutability between the two inputs is governed by the elasticity of substitution $\sigma = 1/(1 - \rho)$. The two labor inputs are perfect substitutes if $\sigma = \infty$. The production function also contains a parameter measuring total factor productivity (A_t) and a productivity parameter measuring the relative efficiency of skilled immigrant and native workers (α):

$$Y_t = R_t[N_t, I_t] = A_t(\alpha N_t^{\rho} + (1 - \alpha)I_t^{\rho})^{\frac{1}{\rho}}$$
(2)

Firms face costs for hiring skilled workers and net costs for training unskilled workers. Both net training costs and hiring costs are quadratic, a mathematical requirement for a closed-form solution. Following Stevens (1994), we assume that the net training costs depend on the number of unskilled workers in training: $C_t[T_t] = \frac{1}{2}c_tT_t^2$. Net training costs are positive $c_t \ge 0$, which implies that training is never profitable per se.⁹ As is plausible in the empirical context studied—where professional associations provide wage recommendations for trainees from which few firms deviate strongly—trainees' compensation is set exogenously and is part of the training costs.

The hiring costs are increasing in the number of hires of skilled workers X_t . We also assume that they decrease in the total supply of skilled labor in the labor market $L^s = L^N + L^I$. The assumption reflects that it is easier and faster to find workers with the required skills if the supply of skilled labor is larger, which lowers their hiring costs (as in Chassamboulli and Peri, 2020). Total hiring costs are thus expressed as $B_t[X_t] = \frac{1}{2}b_t[L^s]X_t^2$, where $b_t \ge 0$ and $b'_t[L^s] < 0$. In support of these assumptions, Blatter et al. (2012), Blatter et al. (2016), and Muehlemann and Strupler Leiser (2018) show that hiring costs for skilled workers in our context are substantial, convex, and increasing in labor market tightness.

⁸Ottaviano and Peri (2012) show that both native and immigrant workers with similar observable skills in the U.S. are imperfect substitutes. Gerfin and Kaiser (2010) provide similar results for Switzerland.

⁹This assumption rules out situations in which firms use training as a means of producing their output more cheaply (the so-called production approach to training, Lindley, 1975). However, the assumption accommodates situations where unskilled workers in training make up for a certain fraction of their training costs near the end of the training period. Studies on the costs and benefits of apprenticeship training in Switzerland suggest that investment and production motives are both important determinants of whether a firm participates in training apprentices (Strupler and Wolter, 2012).

In the first period, firms gain revenue from the production of their skilled workers, pay wages to these workers, incur training costs for unskilled workers, and have hiring costs for new skilled workers. Wages of both types of skilled workers, w_N and w_I , are set competitively. We assume that hiring costs are the same for skilled native and skilled immigrant workers, or that possible differences in hiring costs between the two skilled labor inputs are fully compensated by differences in wage levels. In the second period, firms incur hiring costs if they hire skilled workers. The profit-maximizing firms operate on a competitive product market. Thus, the profit optimization problem is:

$$\max_{T_0, N_0, I_0} A_0 (\alpha N_0^{\rho} + (1 - \alpha) I_0^{\rho})^{\frac{1}{\rho}} - w_0^N N_0 - w_0^I I_0 - \frac{1}{2} c_0 T_0^2 - \frac{1}{2} b_0 [L^s] (N_0 + I_0)^2
+ A_1 (\alpha N_1^{\rho} + (1 - \alpha) I_1^{\rho})^{\frac{1}{\rho}} - w_1^N N_1 - w_1^I I_1 - \frac{1}{2} b_1 [L^s] X_1^2 \quad (3)$$

where, in the first period, all skilled workers are new hires (i.e., $X_0 = L_0$). In the second period, the newly hired skilled workers fill the gap between the required skilled workers and the remaining skilled workers from the first period, together with the trained unskilled workers from the first period: $X_1 = (N_1 + I_1) - (1 - \delta)(N_0 + I_0) - (1 - \gamma)T_0$.

We solve the simple model in appendix A. Combining the first-order conditions for skilled native workers, skilled immigrant workers, and trainees shows that firms train unskilled workers to cover a fraction of the additional skilled workers needed in the future. In general, firms' training depends on the current and future employment of native and immigrant workers and the relative wages between skilled immigrant and skilled native workers. It also depends on the training costs relative to the hiring costs: the greater the training costs relative to the hiring costs, the larger the number of trainees. Firms do not train when the hiring costs approach zero. Conversely, when training costs are zero, firms train all additional skilled workers needed in period 1.

In this model, the exogenous variables affect the decision to train workers in period 0 as follows (see Stevens, 1994, for an extended discussion):

$$T_0 = T\left(b_1[L^I + L^N]^+, c_0^-, \gamma^-, \delta^+ \alpha^-, \frac{w_0^N(L^I, L^N)}{w_0^I(L^I, L^N)}^+\right)$$
(4)

Therefore, the model suggests that a market-level increase in the supply of

skilled immigrant workers—an increase in L^{I} —has two opposing effects on firms' willingness to train. On the one hand, it makes hiring additional skilled workers cheaper in period 1. This effect reduces the firms' willingness to train in period 0 (cost effect). The size of this effect depends on the extent to which the greater supply of skilled immigrant workers affects hiring costs. On the other hand, an increase in L^{I} shifts the market-level labor supply curve of immigrants to the right, leading to a decrease in w_0^{I} .¹⁰ The reduction in wage costs induces firms to hire more workers overall and incentivizes them to train more unskilled workers. This scale effect depends on the degree of complementarity between skilled immigrant and skilled native workers (i.e., σ).

In sum, the simple framework predicts that if skilled native and skilled immigrant workers are perfect substitutes ($\sigma = \infty$), an increase in the availability of skilled immigrant workers decreases the firms' willingness to train because of a reduction in future hiring costs. If skilled immigrants and natives are imperfect substitutes, the effect is unclear and depends on the strength of the cost and scale effects.

3 Institutional background

3.1 Agreement on the free movement of persons

In 1999, Switzerland and the EU signed a bilateral agreement, called the "Agreement on the Free Movement of Person" (hereafter, free movement policy), that introduced free worker mobility for Swiss and EU citizens in Switzerland and the EU. The policy lifted all labor market and immigration restrictions against permanent resident immigrants and cross-border workers from EU countries. In the following, we focus on the liberalizations for cross-border workers because they are the reason why the opening of the labor market had a larger impact on regions close to the border.

There were several hurdles to hiring cross-border workers before the free movement policy came into force. An important restriction was the so-called priority requirement: a firm that wanted to hire a cross-border worker had to provide formal evidence that there was no Swiss worker with skills equivalent to those of the

¹⁰We leave this channel implicit in the model. One approach to model it explicitly is to postulate aggregate labor supply schedules for skilled native and skilled immigrant workers that are increasing in the respective market-level wage. At the labor market level, a greater supply of skilled immigrant workers leads to a reduction in their relative wage (see, e.g., Borjas, 2003).

cross-border worker. This priority requirement was enforced through a governmentcontrolled admission process. Firms had to prepare an application detailing the job requirements, the contract and working conditions, and had to demonstrate that they had searched unsuccessfully in Switzerland for a certain number of weeks. This process took one to three months.

The employment of a cross-border worker was subject to further conditions. A permission to work as a cross-border worker was bound to the specific job, valid for one year only, and granted only to individuals that had lived in a municipality close to the Swiss border for at least six months. Cross-border workers were also required to commute home daily. Moreover, the hiring firm on the Swiss side had to be situated within the so-called "border region", a set of well-defined Swiss municipalities located in vicinity of the Swiss border. Figure 1 shows the border region in Switzerland and illustrates the estimated travel time to the closest border crossing. The border region had been defined bilaterally with each neighboring country between 1928 and 1973 in specific agreements. It is specific to these contracts—it does not follow any cultural, religious, or administrative borders.

The free movement policy abolished these restrictions in a stepwise process. The first legal liberalization step occurred in June 2002 and removed the geographical and occupational restrictions for cross-border workers. Swiss firms in the border regions could now hire anybody from the neighboring countries—the hiring zone was no longer restricted to the border region on the other side of the border. Similarly, cross-border workers were no longer required to go back home daily and to have lived within the border region for at least six months. Finally, their working permits were now valid for five years at every employer. Hence, both new and incumbent cross-border workers now enjoyed unrestricted job mobility within Switzerland. The second legal reform step happened in June 2004. It abolished the priority requirement and the bureaucratic admission process that came along with it. The final reform step in 2007 abolished the "border region". From 2007 onward, all firms in Switzerland gained permission to hire cross-border workers. By implication, the liberalizations in 2002 and 2004 only applied to firms in the border region.

Together, lifting these restrictions caused an unprecedented growth in the employment of cross-border workers. Likely because of natural limits in workers' willingness to commute, this growth was concentrated in Swiss firms within approximately 30 minutes of travel distance to the border. As we show in Tables C.1 and C.2 in the appendix, many of the new cross-border workers had a tertiary degree,



Figure 1: The Swiss border region by travel time to the closest border crossing

Source: Beerli et al. (2021)

Notes: This figure shows the municipalities belonging to the Swiss border region by car travel times to the nearest border crossing. The regions belonging to the non-border regions are in white. The cantonal borders are shown with black, municipal border with grey lines. A canton is a sub-regional entity similar to U.S. states.

took up jobs in skilled occupations, and, in terms of economic sectors, began to work in manufacturing, IT, R&D, business services, real estate, and in the health sector. As a consequence, certain occupations with a high share of apprentices in employment experienced significant increases in the employment share of crossborder workers (see Appendix Figure C.1). Examples include clerical occupations, machine operators, occupations involving medical and social tasks, and occupations in the construction and retail sectors. Apprenticeship graduates were thus exposed to the free movement policy despite the high formal qualifications of the new cross-border workers.

We analyze the effects of the policy by following Beerli et al. (2021)'s transparent Difference-in-Differences approach. We assign firms in Switzerland to one of four groups: Firms are *highly treated* if they are located within 15 minutes of the border within the "border region", *slightly treated* if they are located 15–30 minutes of the border in the border region, and *control* firms if they are either located within *or* outside of the border region and located more than 30 minutes from the border. Since neither control group is clearly more similar to the highly treated group in terms of observables, we generally pool the two control groups. We also show the most important results for the two possible control groups separately. This strategy does not leverage the 2007 abolition of the border region because the 2007 liberalizations affected very few firms.¹¹

Three further comments on the institutional setting are noteworthy. First, we allow for anticipation effects after the policy's approval by the Swiss parliament. Indeed, Beerli et al. (2021) find evidence that certain cantonal offices started to handle cross-border worker applications in a more relaxed way because it became clear that cross-border workers' labor market access would be liberalized eventually. Second, we focus on the reduced-form effects of the policy throughout because we view it as an exogenous increase in the availability of cross-border workers. Hence, we do not present instrumental variable estimates that use the policy to instrument for the cross-border worker or immigrant share in a region. The reason is that the policy likely had effects on firms beyond those mediated through the increased employment of foreigners only.¹² Third, although the free movement policy allowed Swiss cross-border workers to work in Switzerland's neighboring countries, we argue that our estimates reflect the effects of a greater availability of foreign workers to Swiss firms. The reason is that cross-border commuting out of Switzerland remained almost negligible despite a lack of restrictions, probably because it is financially unattractive.¹³ For the same reason, we view it as unlikely that the policy affected training decisions of Swiss firms because it increased the fear that foreign firms may poach the trained workers.

¹¹Almost all firms in the non-border region experienced negligible increases in employment of cross-border workers even after 2007, most likely because most of them are just located too far away from the border (i.e., more than 30 minutes) to attract them (see panel A of Figure 2 below). We discard the few firms located in the non-border region that are located within less than 30 minutes to the border to avoid any confounding from the 2007 policy change.

¹²For instance, the policy increased the geographical and occupational mobility of the preexisting stock of cross-border workers. It also plausibly reduced firms' hiring costs directly by abolishing the priority requirement and by improving firms' chances to find skilled workers. These effects would invalidate the exclusion restriction of the instrumental variable strategy.

¹³Both nominal wages and the cost of living are substantially higher in Switzerland. Therefore, the influx of foreign cross-border workers into Switzerland was nine times higher than the influx of Swiss cross-border workers into its neighbor countries (Beerli et al., 2021).

3.2 Vocational education and training in Switzerland

In Switzerland, vocational education and training (VET) is the largest program at the upper-secondary education level. Every year, about two thirds of a cohort ending compulsory education start a VET track. Nine out of ten VET students take a dual VET track, where they spend about one to two days at a vocational school, obtaining formal education, and three to four days at a training firm, where they learn practical skills and acquire work experience. VET differs from active labor market programs or on-the-job training because it is better integrated into the educational system and more regulated. The federal act on vocational and professional education and training mandates the involvement of the federal government, professional organizations, and cantons and defines their roles (SERI, 2020).

Firms decide on their own whether to participate in training apprentices providing apprenticeships is neither mandatory nor subsidized (Wolter et al., 2006).¹⁴ The programs last between three and four years. In this period, students follow a structured national curriculum, culminating in a final external examination that leads to a national diploma. The national diploma seeks to ensure that graduates learned a defined set of skills in each occupation. Since these skills are transferable between firms within an occupation, trained apprentices are typically considered skilled workers (Mueller and Schweri, 2015). Dionisius et al. (2009) estimate that Swiss apprentices, by their final year, reach 75% of the productivity of an average skilled worker at a firm.

4 Empirical strategy

4.1 Data

Our empirical analyses are based on an establishment-level and a firm-level dataset. The first source is the Swiss business censuses from 1995, 1998, 2001, 2005, and 2008. The censuses provide information on the total employment and workforce composition, including the number of apprentices, of all private and public establishments in Switzerland in September. The censuses also contain the precise geo-coordinates

¹⁴As long as firms comply with the regulations in the federal act on vocational and professional education and training, they may also freely decide in which of the 240 occupations they offer an apprenticeship and to whom. Schweri et al. (2003) present an overview of the factors influencing the costs and benefits of apprenticeship training in Swiss firms.

of each establishment. We use this information to compute the travel time by car to the nearest border crossing.¹⁵ We assign each establishment to its 1998 location throughout the whole sample to avoid biases from endogenous relocation of businesses.

We impose three sample restrictions: we drop establishments from the agricultural sector, which was not covered in all waves, a small number of establishments that we cannot assign to the border or non-border region with certainty, and establishments that did not exist in 1998 when the free movement policy was announced. In our preferred specifications, we additionally focus on a fully balanced sample of firms that existed throughout 1995–2008. Balancing the sample ensures that compositional effects do not drive our results. These effects could arise because the free movement policy led to the entry of new firms in the highly treated region, as shown by Beerli et al. (2021).

Our second dataset is three waves of an administrative and representative firmlevel survey on the costs and benefits of apprenticeship training in Switzerland. These cross-sectional surveys were conducted in 2000, 2004, and 2009. They contain unusually detailed information on firms' motives for training apprentices. We drop firms reporting in training occupations that are either available only in 2004 and 2009 or unavailable for non-training firms.¹⁶ We also drop observations if we are unsure about their assignment to the border or non-border region. We end up with 5,231 observations for non-training firms and 3,837 observations for training firms.

An important disadvantage of the cost-benefit surveys is that we cannot link firms across surveys. An important advantage is that they provide direct evidence of whether firms trade off apprenticeship training against hiring skilled workers externally. Firms were asked to assess whether they train to attract skilled workers, save hiring costs for external workers, save adjustment costs, or secure a talent pipeline in the sector/region. Table C.3 in the appendix provides the list of training motives levied in the surveys and an English translation of the exact survey questions. A nice feature of these data is that firms that did not train were asked to answer these questions in a hypothetical fashion, too. If firms train apprentices in several occupations, the surveys asked firms to answer the questions with reference

¹⁵We use the same data as Beerli et al. (2021) on the location of border crossings in Switzerland to compute the travel duration to the nearest border crossing.

¹⁶The occupations are auto mechanic, carpenter, dental assistant, electronics technician, health specialist, logistics, medical practice assistant, painter, plumber, other three-year VET programs, and other four-year VET programs (in total, 3,663 observations).

to a chosen training occupation.

Table 1 provides descriptive statistics for the two datasets and compares the prereform characteristics of establishments in the four regions that we will compare in our DiD estimation: establishments within the border region up to 15 minutes of the border, establishments within the border region 15–30 minutes of the border, and establishments more than 30 minutes of the border within the border region and the non-border region. Panel A of the table shows that establishments in the highly treated region are somewhat larger than those in the two control groups. Roughly one of four establishments engages in apprenticeship training (24.8%). The average number of apprentices per establishment is 0.65, or 5.5% of total full-time equivalent employment. Among training firms, the apprentice share is 22.3%. As expected, the employment of cross-border workers is concentrated close to the border. A striking fact to which we return later is that firms close to the border are substantially less likely to train apprentices but more likely to employ foreign workers than firms farther away from the border. Turning to the cost-benefit data (panel C of Table 1), we see that the three most important motives for training apprentices are to qualify junior staff into skilled workers, to attract skilled workers, and to secure skilled workers in the sector/region.

Table 1 also suggests that the four regions are quite comparable in a number of important characteristics such as the industry composition. Beerli et al. (2021) additionally show that the regions are also similar in terms of labor market size, a few important worker characteristics, and workers' mean log hourly wages. But we also observe some relevant pre-treatment differences across regions. For instance, highly treated establishments employ more foreigners and are more likely to be exporters and importers. Given the differences in pre-treatment establishment characteristics, we will probe the robustness of our results if we use covariate matching to generate a control group comparable to highly treated establishments in terms of these and other pre-treatment characteristics.

4.2 Regression model

We estimate the effect of the greater availability of skilled workers on firms' training behavior using the Difference-in-Differences (DiD) design by Beerli et al. (2021). The approach leverages the larger impact of lifting restrictions for cross-border workers on establishments near the border. We assign firms to one of three groups:

			Borde	r region			Non-	border
Travel time to border	<u><</u> 1	5 min	15-	30 min	>30) min	\mathbf{re}	gion
	Mean	SD	Mean	SD	Mean	$^{\mathrm{SD}}$	Mean	SD
A. Business census (all)								
FTE employment	11.46	(57.24)	12.14	(78.40)	9.53	(43.41)	9.89	(38.44)
Foreign share (in FTE, 1995)	0.24	(0.32)	0.16	(0.27)	0.14	(0.25)	0.10	(0.21)
Cross-border worker share (1995)	0.08	(0.17)	0.01	(0.05)	0.00	(0.04)	0.00	(0.01)
Share of other foreigners (1995)	0.17	(0.27)	0.15	(0.25)	0.13	(0.24)	0.10	(0.21)
Training firm $(0/1)$	0.21	(0.41)	0.25	(0.43)	0.25	(0.43)	0.28	(0.45)
Apprentice share	0.04	(0.11)	0.05	(0.12)	0.06	(0.13)	0.06	(0.13)
Manufacturer $(0/1)$	0.12	(0.32)	0.11	(0.32)	0.12	(0.33)	0.13	(0.33)
High-tech manufacturer $(0/1)$	0.03	(0.18)	0.03	(0.17)	0.03	(0.18)	0.03	(0.17)
Construction firm $(0/1)$	0.09	(0.28)	0.09	(0.29)	0.10	(0.30)	0.11	(0.31)
Publicly owned firm $(0/1)$	0.12	(0.33)	0.13	(0.33)	0.14	(0.34)	0.14	(0.35)
Exporter $(0/1, 1995)$	0.16	(0.37)	0.15	(0.35)	0.12	(0.33)	0.11	(0.31)
Importer $(0/1, 1995)$	0.26	(0.44)	0.26	(0.44)	0.22	(0.41)	0.21	(0.41)
Travel minutes to border	7.13	(3.54)	23.36	(4.07)	39.25	(9.92)	53.65	(13.57)
Observations	42623	()	55500	()	26905	()	55947	()
B. Business census (training only)								
FTE employment	27.59	(110.79)	27.59	(150.66)	21.34	(80.85)	20.79	(68.54)
Foreign share (in FTE)	0.24	(0.26)	0.16	(0.22)	0.13	(0.20)	0.09	(0.17)
Apprentice share	0.21	(0.15)	0.22	(0.15)	0.23	(0.15)	0.23	(0.15)
Observations	8907	()	13718	()	6774	()	15468	()
C. Cost-benefit surveys								
Training firm	0.14	(0.35)	0.15	(0.36)	0.20	(0.40)	0.19	(0.39)
Number of apprentices	0.34	(1.56)	0.41	(2.14)	0.46	(1.64)	0.44	(1.42)
Apprentice share	0.03	(0.15)	0.03	(0.12)	0.05	(0.13)	0.05	(0.15)
Attract skilled workers	3.82	(1.24)	3.86	(1.18)	3.75	(1.19)	3.89	(1.16)
Hiring cost	2.69	(1.15)	2.62	(1.12)	2.50	(1.13)	2.56	(1.12)
Risk of wrong decision	3.01	(1.22)	2.87	(1.20)	2.78	(1.21)	2.83	(1.17)
Replace unskilled workers	2.62	(1.17)	2.69	(1.22)	2.59	(1.24)	2.54	(1.17)
Adjustment cost	2.84	(1.25)	2.79	(1.22)	2.74	(1.19)	2.77	(1.15)
Hire the best	3.19	(1.24)	3.14	(1.24)	3.04	(1.26)	3.07	(1.19)
Avoid fluctuation	3.25	(1.23)	3.20	(1.16)	3.05	(1.20)	3.06	(1.16)
Qualify junior staff	3.84	(1.15)	3.78	(1.10)	3.67	(1.14)	3.71	(1.11)
Secure skilled workers	3.72	(1.23)	3.81	(1.23)	3.80	(1.21)	3.87	(1.17)
Firm size	16.90	(57.41)	15.81	(60.62)	11.66	(43.85)	12.95	(38.84)
Observations	725	()	1074	()	560	(951	()

Table 1: Firm characteristics before the reform, by region and distance to the border

Notes: The table shows descriptive statistics of establishments in the border and non-border region using the business census (BC, panels A and B) in 1998 (unless noted otherwise) and firm characteristics according to the cost-benefit survey in 2000 (panel C). The border region is split into groups depending on firms' travel duration to the nearest border crossing. The data in panel A is restricted to establishments existing throughout 1995–2008. The "share of other foreigners" encompasses all non-Swiss workers that are not cross-border workers. Panel B focuses on establishments that train apprentices. Panel C shows descriptive statistics using the cost-benefit data in 2000. The training motives are measured on a 5-point Likert scale. The table shows the mean of the ordinal variable. Table C.3 provides the list of training motives levied in the surveys and an English translation of the exact survey questions.

establishments located within 15 minutes travel time to the nearest border crossing d_i within the border region, $I(d_i \leq 15)$, establishments located 15 to 30 minutes of the border within the border region, $I(15 < d_i \leq 30)$, and establishments more than 30 minutes of the border either within or outside the border region. We then interact these indicators with indicators of the survey years t. For the business censuses that took place in 1995, 1998, 2001, 2005, and 2008, we estimate variants of the following DiD model:

$$y_{i,t} = \beta_{high}^{t=2001} [I(d_i \le 15) * I(t = 2001)] + \beta_{high}^{t\ge 2005} [I(d_i \le 15) * I(t \ge 2005)] + \beta_{slight}^{t=2001} [I(15 < d_i \le 30) * I(t = 2001)] + \beta_{slight}^{t\ge 2005} [I(15 < d_i \le 30) * I(t \ge 2005)] + \alpha_i + \alpha_t + \epsilon_{i,t}$$

$$(5)$$

In this model, $y_{i,t}$, represents the training or employment outcome of firm i in year t. The β coefficients capture the DiD of this outcome for highly treated firms (β_{high}) and slightly treated (β_{slight}) firms *relative* to control establishments. For both slightly and highly treated establishments, we estimate separate effects for the year 2001, to capture anticipation effects of the reform, and for the years 2005 and 2008 to capture an average impact one and four years after full liberalization. The model controls for year fixed effects (α_t) , which capture aggregate macroeconomic shocks common to all firms such as changes in aggregate prices and foreign demand, and establishments fixed effects (α_i) , which control for the baseline effects of establishments' distance to the border (d_i) and other pre-existing differences between establishments and regions.

The central identifying assumption in our research design is that establishments in the three regions would have, on average, had the same *change* in outcomes had the border not opened. This common trend assumption is violated, for instance, if there are unobserved third factors that affect the training behavior of firms near the border differently from firms farther away at the time of the border opening. Potential confounders are simultaneous, region-specific policy changes or shocks to prices, demand, or productivity that have region-specific effects because of differences in the sectoral composition between regions.

While we cannot test the validity of the common trend assumption, we can assess its plausibility by assessing pre-treatment trends in outcomes. Thus, we also estimate an event study version of equation 5 that contains year-specific effects for each census year except 1998 that serves as the reference year. The event study allows us to examine whether establishments near and farther from the border displayed similar trends in training outcomes before the free movement policy. Admittedly, however, our possibilities to assess pre-trends are somewhat limited since our datasets do not cover a long pre-treatment period. Against this background, it is reassuring that Beerli et al. (2021), and the various follow-up papers that use the same research design (Ariu, 2020; Bächli and Tsankova, 2020; Cristelli and Lissoni, 2020; Naguib, 2019), present evidence of common pre-trends for a variety of firm outcomes in our setting, including firm size, productivity, wages, worker composition, innovation, and patents, in some cases with data that goes back until the 1980s. In addition, section 5.4 presents several robustness checks that probe the robustness of our results if we control for more restrictive sets of fixed effects such as industry times year or labor market region times year fixed effects. We also present DiD estimates that use a matched control group similar to treated units in terms of several pre-treatment observables, including industry affiliation, international exposure, and firm size. This approach is robust to nationwide shocks that affect firms differently along these or correlated dimensions.

5 Results

This section explores the causal effect of a greater availability of skilled workers on the establishments' willingness to train unskilled workers.

5.1 Descriptive evidence

Figure 2 shows the employment of cross-border workers and training provision separately for highly treated establishments, slightly treated establishments, and establishments in the two control groups. Panel (a) uses the censuses in 1995 and 2008, which contain information on establishments' employment of cross-border workers, to show the share of cross-border workers in total full-time equivalent employment. The figure demonstrates that employment of cross-border workers is highly concentrated near the border, both before and after the free movement of workers. In 2008, cross-border workers made up more than a sixth of the workforce in border region establishments within 15 minutes to the border. The figure also shows that the growth in the employment share of cross-border workers between 1995 and 2008 is substantial and concentrated close to the border, too.

Panel (b) of Figure 2 shows the fraction of establishments engaged in the training of apprentices. Panel (c) shows the percentage of apprentices in total employment. The two figures demonstrate that establishments near the border offer remarkably fewer training positions than establishments farther away. The magnitudes are considerable: the share of apprentices in total employment is 34% lower in establishments within 15 minutes compared to those located at least 30 minutes away from the border. Similarly, the share of firms that train at least one apprentice (panel b) is 24% lower.

Taken together, Figure 2 shows that employment of cross-border workers and apprenticeship training are inversely related in the cross-section, suggesting that the two are substitutes. Aepli and Kuhn (2021) argue based on this cross-sectional relationship that cross-border workers displace apprentices. However, we do not observe a strong relationship between distance to the border and the over-time *changes* in the training and apprentice share in panels b) and c) of Figure 2 although there is a simultaneous growth in cross-border employment close to the border. It may thus be that other differences between firms near and farther from the border explain why firms near the border train less. For example, firms close to the border are more likely to be importers and exporters than firms farther away (see Table 1). Swiss firms with international exposure have a lower propensity to train apprentices (Muehlemann, 2014). We thus now present DiD estimates that account for timeinvariant differences between firms' propensity to train as a response to a large, exogenous increase in the availability of cross-border workers.

5.2 Effect on the employment of foreign workers

Using data from the business census 1995–2008, Figure 3 shows estimates of the effect of the free movement policy on the employment of foreign workers in highly and slightly treated establishments. The figure shows the results of an event study version of equation (5) that separately estimates the policy effects for each census year. The dependent variable is full-time equivalent employment of foreign workers—cross-border workers plus resident workers without a Swiss passport—relative to establishments' total full-time equivalent employment in 1998, when the policy was announced. This outcome retains firms without foreign workers. Because



Figure 2: Employment of cross-border workers and apprentices by region

(a) Cross-border worker share

(b) Training share

Notes: This figure shows establishments' cross-border worker share and training provision by travel distance in minutes to the nearest border crossing. The figure uses data from the business censuses. We differentiate highly treated establishments (establishments within 15 minutes of the border, termed "BR, <15 min"), slightly treated establishments (establishments within 15–30 minutes of the border, "BR, 15–30 min"), establishments more than 30 minutes of the border within the border region ("BR, 30+ min"), and establishments in the nonborder region ("NBR, 30+ min"). Panel (a) shows the employment percentage of cross-border workers. We use 1995 data because the 1998 census provides no information on cross-border workers. Panel (b) shows the fraction of establishments that train apprentices. Panel (c) shows the employment percentage of apprentices. The fractions in panels (a) and (c) are employment-weighted. The black lines show the standard errors of the sample means.

we hold the denominator fix, it also separates an effect on foreign employment from a possible effect on firm size. We winsorize the outcome at the top 0.01% value to reduce the influence of very few extreme outliers. We use our preferred sample: all establishments that exist throughout 1995–2008 (see section 4.1). To allow for arbitrary dependence between units within the same commuting zone (both cross-sectional dependence and over time), we cluster standard errors at the level of commuting zones (NUTS-III regions). Finally, we weight the regression by firm size prior to the policy change.¹⁷ We discuss how these specification choices affect our results in section 5.4.

Figure 3 shows that full-time equivalent employment of foreign workers as a percentage of 1998 employment grows consistently more in highly treated establishments after 1998 than establishments in the control regions. By 2008, the excess increase amounts to approximately six percentage points. Since the average highly treated establishment employed 11.46 FTE workers in 1998 (see Table 1), foreign employment grew by roughly 6% * 11.46 * 42'623 = 29'100 foreigners more in the 42'623 highly treated establishments compared to the establishments in the two control groups. As expected, we find a smaller impact on slightly treated establishments—those located between 15–30 minutes to the border. Importantly, none of the placebo effects for the 1995–1998 period is significantly different from zero, suggesting similar trends in foreign workers' employment in the treatment and control groups in this period.

Panel A of Table 2 uses the same outcome but the simpler DiD model (equation (5)) to provide two additional insights. The main coefficient of interest is the interaction term $I(t \ge 2005) \cdot I(d_i \le 15)$ that captures the policy effects on highly treated establishments after cross-border workers had unrestricted access to the border region. The table, first, shows that the impact on highly treated establishments is very similar in establishments that employ at least one apprentice throughout 1995–2008 (column 4). Second, the estimated effects are quantitatively similar if we use the two control groups separately (columns 5 and 6).

Overall, these findings confirm that the opening of the Swiss labor market to cross-border workers had a large positive impact on the employment of foreign workers in firms in the border region. The section extends similar results by Beerli et al. (2021) by showing that this effect also pertains to establishments that train apprentices. Estimating similar regression models by educational attainment and occupation, Beerli et al. (2021) show that these additional hirings of foreigners primarily consisted of *skilled*, tertiary-educated workers, consistent with the descriptive patterns in Appendix Table C.2.

¹⁷Initial firm size is a firm's average full-time equivalent employment in the waves 1995 and 1998 of the census. In estimations that incorporate firms founded after 1998, we use a firm's size in the first census wave that the firm appears.

(3)(4)(6)VARIABLES **Ô**LS ÈΈ ÈΈ ÈÉ ÈΈ FE A. Foreign workers / total FTE employment in 1998 $I(t = 2001) \cdot I(d_i \le 15)$ 0.015** 0.022*** 0.021*** 0.012 -0.002 0.014^{*} (0.008)(0.006)(0.008)(0.006)(0.007)(0.007)0.015*** $I(t = 2001) \cdot I(15 < d_i \le 30)$ 0.007 0.014** 0.017^{*} 0.020*** 0.012 (0.005)(0.005)(0.006)(0.009)(0.006)(0.007) $I(t \ge 2005) \cdot I(d_i \le 15)$ 0.031*** 0.043*** 0.041*** 0.045*** 0.046*** 0.042*** (0.011)(0.009)(0.010)(0.011)(0.011)(0.010) $I(t \ge 2005) \cdot I(15 < d_i \le 30)$ 0.012* 0.015** 0.015^{*} 0.016^{*} 0.018** 0.014^{*} (0.007)(0.007)(0.007)(0.006)(0.010)(0.009)Observations 1,442,654 904,900 904,900 345,080 625,140770,375 Mean dep. variable in 1998 .153.149.149.158.174.153**B.** Apprentices / total FTE employment in 1998 $I(t = 2001) \cdot I(d_i \le 15)$ 0.003 -0.001 -0.000 -0.000 0.000 -0.001 (0.002)(0.002)(0.002)(0.003)(0.002)(0.002) $I(t = 2001) \cdot I(15 < d_i \le 30)$ 0.001 -0.000 -0.001 -0.000 0.001 -0.001 (0.001)(0.002)(0.002)(0.002)(0.001)(0.002) $I(t \ge 2005) \cdot I(d_i \le 15)$ -0.005* -0.004-0.002-0.005 -0.003 -0.004(0.004)(0.002)(0.003)(0.002)(0.003)(0.003) $I(t \ge 2005) \cdot I(15 < d_i \le 30)$ -0.000 -0.000-0.000 -0.000 0.001 -0.001(0.002)(0.002)(0.002)(0.003)(0.003)(0.002)Observations 1,442,654 904,900 904,900 345,080 625,140 770,375 Mean dep. variable in 1998 .057.061 .044 .061.061.161 C. Training provision (0/1) $I(t = 2001) \cdot I(d_i \le 15)$ -0.000 -0.006* -0.006 -0.006 -0.006* -0.006 (0.003)(0.009)(0.004)(0.002)(0.004)(0.004) $I(t = 2001) \cdot I(15 < d_i \le 30)$ 0.0030.002 0.0020.0070.0020.002(0.002)(0.002)(0.002)(0.006)(0.003)(0.002) $I(t \ge 2005) \cdot I(d_i \le 15)$ -0.006-0.007-0.006-0.007-0.007-0.007(0.005)(0.006)(0.005)(0.014)(0.006)(0.006) $I(t \ge 2005) \cdot I(15 < d_i \le 30)$ 0.003 0.004 0.004 0.014* 0.004 0.004 (0.003)(0.003)(0.003)(0.007)(0.003)(0.003)1,878,844 904,900 904,900 345,080 625,140 770,375 Observations Mean dep. variable in 1998 .172 .247 .247 .650 .235 .247 Control group Both Both Both Both BR 30+NBR Balanced sample Yes Yes Yes No Yes Yes Period FE Yes Yes Yes Yes Yes Yes Region FE Yes Yes Yes Yes Yes Yes Establishment FE No Yes Yes Yes Yes Yes Apprentice supply control No No Yes No No No Training establ. only No No No Yes No No

Table 2: Effect of free movement policy on foreign employment and apprenticeships

Notes: This table shows the effect of the free movement policy on foreign employment and the provision of apprenticeships using our main DiD model (equation (5)). The regressions are based on the business censuses (BC) 1995–2008. The dependent variable in panel A is full-time equivalent (FTE) employment of foreign workers relative to total full-time equivalent employment in 1998. The dependent variable in panel B is the number of apprentices relative to total full-time equivalent employment in 1998. The dependent variable in panel C is a dummy variable equal to 1 if an establishment trains apprentices. The main coefficient of interest is the interaction between firms within 15 minutes commuting time d_i to the border and the free movement period, $I(t \ge 2005) \cdot I(d_i \le 15)$. The control group in columns 1-4 is establishments located in the border region with more than 30 minutes travel time to the border (BR 30+) and establishments in the non-border region (NBR). Results for each control group separately are provided in columns 5 and 6. The sample in column 1 is all establishments in the BC. The "balanced sample" used in the remaining columns comprises of all establishments existing throughout 1995–2008. The sample in columns 4 is additionally restricted to establishments that train at least once throughout 1995–2008. Regressions in panels A and B are weighted using establishments' average employment 1995 and 1998 as weight. The "apprentice supply control" used in column 3 is the number of graduates to from lower secondary schools (11th grade) in the commuting zone (NUTS-III region). Standard errors are clustered on the level of commuting zones. ***, **, *, denote statistical significance at the 1%, 5%, and 10% level, respectively.



Figure 3: Timing of effect of free movement policy on foreign employment

Notes: The figure shows the effect of the free movement policy on establishments' employment of foreign workers using data from the Swiss business censuses 1995–2008. It plots the estimated policy effects and associated 95% confidence intervals using a generalization of our main DiD model (equation (5)) that estimates separate effects for each census year. We standardize the effects to 0 in 1998 by dropping the indicator for that year from the regression. The estimation sample comprises of all establishments existing throughout 1995–2008. The dependent variable is full-time equivalent employment of foreign workers (cross-border workers plus foreign resident immigrant workers) relative to total full-time equivalent employment in 1998. We estimate effects for all highly treated establishments (within 15 minutes of the border, termed "BR, 0–15 min")) and slightly treated establishments (within 15–30 minutes of the border, termed "BR, 15–30 min"). The control group is establishments located more than 30 minutes away from the border. The regression is weighted using establishments' average employment pre-1999 as weight. We control for establishment and period fixed effects. Confidence intervals are clustered on the level of commuting zones.

5.3 Effect on the firms' provision of training

Did the free movement policy, and the growth in employment of skilled foreign workers in training firms to which it led, affect the number of apprenticeships that establishments offer? We present our main DiD estimates based on the censuses 1995–2008 in Panel B of Table 2. Figure 4 shows the corresponding event study. The outcome variable is the number of apprentices trained in an establishment relative to total employment in 1998, consistent with the specification of the outcome in the previous section. As before, we winsorize the outcome and weight observations using establishments' pre-1999 employment.

The event study in Figure 4 does not suggest that the greater availability of skilled foreign workers affected firms' provision of apprenticeships. Relative to the two control groups, apprenticeships began to decline in highly treated establishments after 1998. By 2008, the negative point estimate amounts to 0.4 percent



Figure 4: Timing of effect of free movement policy on apprenticeship training

Notes: The figure shows the effect of the free movement policy on the establishments' provision of apprenticeships using data from the Swiss business censuses 1995–2008. It plots the estimated policy effects and associated 95% confidence intervals using a generalization of our main DiD model (equation (5)) that estimates separate effects for each census year. We standardize the effects to 0 in 1998 by dropping the indicator for that year from the regression. The estimation sample comprises of all establishments existing throughout 1995–2008. The dependent variable is the number of apprentices relative to total full-time equivalent employment in 1998. We estimate effects for highly treated establishments (within 15 minutes of the border, "BR, 0–15 min") and slightly treated establishments located more than 30 minutes away from the border. The regression accounts for establishment and period fixed effects and is weighted using establishments' average employment pre-1999 as weight. Confidence intervals are clustered on the level of commuting zones.

of total 1998 full-time equivalent employment. However, the policy effects are not statistically significant at conventional levels. Moreover, in contrast to the sharp increase in the employment of foreign workers in the same establishments, the negative estimate emerges gradually over an extended period, which makes it hard to attribute the relative decline in apprenticeship training to the free movement policy. The estimated effects are also close to zero and statistically insignificant for slightly treated establishments in all census years.

The DiD estimates that correspond to the event study, presented in panel B of Table 2, confirm that the free movement policy had limited effects on the employment of apprentices. The exception is column 1. The coefficient for highly treated establishment, $I(t \ge 2005) \cdot I(d_i \le 15)$, is statistically significantly negative at the 10 percent significance level in this specification. However, since we include all establishments that exist in 1998 and do not control for establishment fixed effects, these results could be biased downward because of firm entry.¹⁸ All other

¹⁸Beerli et al. (2021) show that the policy led to the entry of firms close to the border. New firms are, on average, smaller than incumbent firms. Smaller firms have a lower probability to train apprentices than larger firms (Muehlemann and Wolter, 2007). The entry of new firms thus likely

specifications in the table produce estimates that are close to zero and statistically insignificant. For instance, we get similar estimates if we absorb possible trend differences in the local supply of apprentices by adding a control variable that reflects the (potential) supply of apprentices in the local labor market.¹⁹ Columns 5 and 6 show that the findings do not depend on the control group either. Finally, there is no evidence that the policy affected the extensive margin—the probability that firms provide at least one training position. These results are shown in panel C of Table 2.²⁰ The dependent variable of these unweighted regressions is a dummy equal to one if an establishment engages in apprenticeship training.

To put the estimated impact of the free movement policy on the number of apprenticeships in perspective, we can compare it to the effect on the hiring of foreign workers. Panel (a) of Figure 5 provides this comparison for highly treated establishments. We observe that the effect on apprentices is an order of magnitude smaller than the effect on foreign employment. Taken at face value, the estimates suggest that each foreign worker hired by highly treated establishments is associated with a decrease of about 0.09 apprenticeship positions. Alternatively, the 0.0037 percentage point effect on highly treated establishments in our main specification implies that the free movement policy reduced the number of apprenticeship positions by 6%, or approximately 1200 positions in total.²¹

Do these aggregate estimates hide heterogeneity between industries? Figure 6 shows separate estimates of our main event study model for broad industries. The figure illustrates that the negative point estimate of our aggregate model is primarily the result of highly treated establishments in the construction and service sectors. Indeed, we find statistically significant evidence at the 5 percent level that the free movement policy reduced apprenticeship provision in the construction sector. There is no statistically significant evidence for a displacement of apprenticeships in the

reduces the percentage of firms that train in the highly treated region. This effect might lead us to overstate a possible negative effect of the free movement policy on the training of apprentices.

¹⁹The control variable measures the number of graduates from lower secondary schools (11th grade) in the commuting zone (NUTS-III region) in a given year.

²⁰Indeed, the estimated effects on *slightly* treated establishments are *positive* in both periods and all specifications, and even statistically significantly so at the 10 percent level if we focus only on firms that train at least once (column 4).

 $^{^{21}}$ Aepli and Kuhn (2021) present similar back-of-the-envelope calculations. They estimate that the growth in cross-border employment between 1995 and 2008 led to about 3500 fewer apprenticeship positions. These estimates are not directly comparable to ours since we quantify the effect of the free movement policy while Aepli and Kuhn (2021) focus on the growth in cross-border workers.

Figure 5: Effect of free movement policy on apprenticeship training: Comparison of effect size



(a) Establishment-level estimates

Notes: The figure shows the effect of the free movement policy on employment of foreign workers and the provision of apprenticeships in highly treated units using data from the Swiss business censuses 1995–2008. It plots the estimated policy effects and associated 95% confidence intervals using a generalization of our main DiD model (equation (5)) that estimates separate effects for each census year. We focus on highly treated establishments (establishments within 15 minutes of the border) and standardize the effects to 0 in 1998 by dropping the indicator for that year from the regression. Panels (a) and (c) are estimated using establishment-level data. The dependent variables are (i) full-time equivalent employment of foreign workers relative to total full-time equivalent employment in 1998, and (ii) the number of apprentices relative to total employment in 1998. Panel (a) shows the results with our baseline approach based on a balanced panel of establishments existing throughout 1995–2008. Panel (b) shows analogous estimates if we use all establishments existing in a given year and aggregate the data on municipal level. Panel (c) shows our baseline establishment-level DiD regressions if we use a control group of non-treated establishments that we match to highly treated establishments using Mahalanobis (covariate) distance matching (see appendix B for details). The regressions are weighted using average employment pre-1999 as weight. In panel c, the weight is additionally multiplied with the number of times that a control establishment is matched to a highly treated establishment. All regressions account for period fixed effects and establishment (panels a and c) or region (panel b) fixed effects, respectively. Confidence intervals are clustered on the level of commuting zones.

three other sectors. In all sectors, highly treated and control establishments display parallel pre-trends in the number of apprentices.

Overall, our evidence suggests that the greater availability of cross-border workers did not reduce firms' willingness to provide apprenticeships. We reach a similar conclusion if we analyze firms' training provision in the cost-benefit surveys.²² An exception is the construction sector where apprenticeship training and access to skilled foreign workers appear to be substitutes. Importantly, the point estimates also rule out (large) positive effects on the number of apprentices: apprentices do not appear to have profited from the free movement policy in contrast to tertiaryqualified workers, which experienced wage and possibly employment growth because of the policy (see Beerli et al., 2021).





Notes: The figure shows the effect of the free movement policy on the establishments' provision of apprenticeships estimated separately by broad economic sector of activity. We use data from the Swiss business censuses 1995–2008. It plots the estimated policy effects and associated 95% confidence intervals using a generalization of our main DiD model (equation (5)) that estimates separate effects for each census year. The DiD estimates based on equation (5) are presented in panel C of Table 4. We standardize the effects to 0 in 1998 by dropping the indicator for that year from the regression. The estimation sample comprises of all establishments existing throughout 1995–2008. The dependent variable of each regression is the number of apprentices relative to total employment in 1998. We focus on highly treated establishments. The control group in all panels is establishments more than 30 minutes from the border. All regressions account for establishment and period fixed effects. The regressions are weighted using establishments' average employment pre-1999 as weight. Confidence intervals are clustered on the level of commuting zones.

²²Indeed, the point estimates presented in appendix table C.5 suggest that the free movement policy if anything increased firms' provision of apprenticeship training.

5.4 Robustness

This section assesses how robust these baseline results are. Table 3 adds more demanding sets of fixed effects to our baseline model and presents the results of a few alternative specifications. The table shows that the estimates are comparable if we control for unobserved industry-specific shocks by including two-digit industry times year fixed effects $(\text{column } 1)^{23}$, and unobserved regional shocks by including NUTS-II region times year (column 2) and canton times year fixed effects (column 3). The latter model is identified only from comparing firms with different distances to the border located within the same of the 26 Swiss cantons. The estimated effects are also similar if we restrict the sample to two-digit industries unaffected by either of the other bilateral agreements introduced along with the free movement policy $(column 4)^{24}$, if we use the somewhat smaller estimation sample and the weighting scheme that Beerli et al. (2021) use (column 5), if we do not weight observations by establishment size (column 6), and if we do not winsorize the outcome variables (column 7). Importantly, the main effect on the apprentice share (panel B) turns statistically significantly negative in columns 2 and 6. Yet, there are no a priori reasons to prefer those estimates over the others. Indeed, our baseline estimates (column 2 of Table 2) are close to the average of the estimates in Tables 2 and 3.

Since we normalize our main outcome variables with firm sine in 1998, all regressions presented so far only use firms that exist in 1998. Our preferred specification is additionally restricted to firms existing throughout 1995–2008. We probe the

 $^{^{23}}$ The robustness to including industry-year fixed effects also limits concerns that our results are affected by a major revision of the federal act on vocational and professional education that took place in 2004. The reform included health, social, art, and agriculture and forestry occupations in the VET system from 2004 onwards (BBT, 2003). Although this reform touched all regions in Switzerland at the same time, it could have region-specific effects because of differences in the occupational composition across regions. Since the reform affected occupations concentrated in certain industries, the industry-year fixed effects likely control for a large part of the possible reform effects. To further assess the robustness of our results to this major reform, we use the costbenefit data and exclude all training occupations that are directly affected by the reform. Table C.6 thus disregards carpenters, computer scientists, electricians, retail specialists, and salesman. The results are similar to our baseline results (Table C.5) with this restriction.

²⁴The free movement policy was part of a package of bilateral agreements between the EU and negotiated at the same time as the agreement on the free movement of persons. One of these agreements, for example, reduced non-tariff barriers to trade between Switzerland and the EU. This trade liberalization may have affected regions near the border more than the regions farther away. Beerli et al. (2021) thus use a proxy for exposure to these other agreements based on a classification by Buehler et al. (2011). Buehler et al. (2011) carefully assess the extent to which a specific two-digit industry was affected by these other agreements. Column 7 of Table 3 is restricted to non-affected two-digit industries.

relevance of these restrictions in Panel (b) of Figure 5 by aggregating employment of *all* establishments in the census to the municipality level. We then rerun our DiD model on this level. Reassuringly, the resulting estimates are very similar to our baseline estimates. As we show in Appendix Table C.4, the municipality-level regressions also provide no evidence that the free movement policy affected the number and the share of establishments that train apprentices.

An important further concern with our results is that there are noteworthy differences in observed characteristics between treated and control firms prior to the free movement policy (see Table 1). These differences raise the possibility that unobserved shocks to these dimensions or dimensions correlated with them confound our estimates. We address this concern using a matching approach that makes treated and control establishments close to identical in several pre-reform characteristics, including those that are imbalanced in Table 1. We then run our DiD regressions using highly treated and control establishments that are similar *ex ante*, limiting concerns that unobserved shocks could affect treatment and control group differently *ex post*.

We use Mahalanobis covariate matching to achieve these goals (see appendix B for details). We match one control establishment to each highly treated establishment. We only match firms in the same two-digit industry and same firm size class that, in addition, agree in terms of indicators whether they trained apprentices prereform, employed foreign workers, and existed in 1991. Moreover, we make firms as similar as possible in terms of some other pre-reform characteristics, including indicators of public ownership and export and import status. We then run our baseline regression with this matched control sample by multiplying the employment weight by the number of times a control establishment is matched to a highly treated establishment. We loose 133 highly treated establishments without an exact match. Panel (c) of Figure 5 presents the results. We observe that the matching estimates confirm our baseline results, presented in panel (a), in terms of sign and size of the effects.

5.5 Mechanisms

The previous sections suggest that a greater availability of skilled foreign workers caused a strong growth in the number of foreign workers but had no or at most a small negative impact on firms' provision of apprenticeship training. Our theoretical

Table 3:	Main	robustness	checks
----------	------	------------	--------

	(1)	(2)	(3)	(4)	(5)	(6)	(7)
	Industry-	NUTS-II-	Canton-	Not exposed	BRSP	No	Including
VARIABLES	period FE	period FE	period FE	to bilaterals	sample	weights	outliers
	A. Foreign	workers / t	otal FTE en	nployment in	1998		
$I(t = 2001) \cdot I(d_i < 15)$	0.011*	0.017^{**}	0.024^{***}	0.021**	0.032^{***}	0.010^{***}	0.016^{*}
	(0.006)	(0.007)	(0.009)	(0.009)	(0.006)	(0.003)	(0.009)
$I(t = 2001) \cdot I(15 < d_i < 30)$	0.014***	0.020***	0.016***	0.021*	0.024***	0.006**	0.017**
	(0.005)	(0.005)	(0.004)	(0.011)	(0.007)	(0.003)	(0.008)
$I(t > 2005) \cdot I(d_i < 15)$	0.035***	0.038***	0.033**	0.039***	0.079***	0.045***	0.045***
	(0.007)	(0.008)	(0.014)	(0.010)	(0.017)	(0.011)	(0.010)
$I(t > 2005) \cdot I(15 < d_i < 30)$	0.015***	0.023***	0.019***	0.012	0.032***	0.014***	0.018**
	(0.005)	(0.006)	(0.006)	(0.012)	(0.010)	(0.003)	(0.008)
	()	()	()	()	()	()	()
	B. Appre	ntices / tot	al FTE emp	loument in 19	98		
$I(t = 2001) \cdot I(d_i < 15)$	-0.001	-0.001	-0.000	0.001	-0.005***	-0.002	0.000
= (1 1 1 1 1 1 1 1 1 1	(0.002)	(0.002)	(0.001)	(0.002)	(0.002)	(0.001)	(0.004)
$I(t = 2001) \cdot I(15 < d_i < 30)$	0.000	-0.001	0.000	0.000	-0.000	0.000	-0.002
$1(b = 2001)$ $1(10 < w_i \le 00)$	(0.000)	(0.001)	(0.000)	(0.002)	(0.001)	(0.000)	(0.002)
$I(t \ge 2005) \cdot I(d_t \le 15)$	-0.003	-0.005***	-0.000	-0.002	-0.005	-0.006**	-0.006
$\Gamma(v \geq 2000) \cap (w_i \leq 10)$	(0.002)	(0.002)	(0.000)	(0.002)	(0.003)	(0.000)	(0.004)
$I(t > 2005) \cdot I(15 < d \le 30)$	0.002)	-0.001	-0.000	-0.000	0.001	0.000	-0.002
$\Gamma(t \geq 2000) \Gamma(10 < \omega_t \leq 00)$	(0.002)	(0.001)	(0.000)	(0.003)	(0.001)	(0.002)	(0.002)
	(0.002)	(0.001)	(0.002)	(0.005)	(0.002)	(0.002)	(0.005)
		C Trainin	a provision	(0/1)			
$I(t = 2001) \cdot I(d_t \le 15)$	-0.006	-0.003	-0.000	-0.004	-0.002		
$1(b = 2001) - 1(a_1 \le 10)$	(0.004)	(0.003)	(0.000)	(0.001)	(0.002)		
$I(t = 2001) \cdot I(15 < d \le 30)$	0.002	0.002	0.003	0.003	0.015		
$I(t=2001) \cap I(10 < u_i \le 50)$	(0.002)	(0.002)	(0.003)	(0.003)	(0.010)		
$I(t > 2005) \cdot I(d_1 < 15)$	-0.005	-0.002)	0.002)	0.001	0.004		
$I(l \ge 2000) \cap (u_l \ge 10)$	(0.006)	(0.003)	(0.003)	(0.001)	(0.004)		
$I(t > 2005) \cdot I(15 < d \le 30)$	0.005*	0.003	0.003	0.007**	0.022*		
$\Gamma(t \geq 2000) \Gamma(10 < \omega_t \leq 00)$	(0.003)	(0.002)	(0.002)	(0.001)	(0.009)		
	(0.000)	(0.002)	(0.002)	(0.000)	(0.000)		
Observations	904,896	904,495	904,900	582,068	441,025	904,900	904,900
Control group	Both	Both	Both	Both	Both	Both	Both
Balanced sample	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Establishment FE	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Period FE	Yes	Yes	Yes	Yes	Yes	Yes	Yes
	1	6 + 1 C			1	• • •	

Notes: This table assesses the robustness of the effect of the free movement policy on foreign employment and apprenticeship training using our main DiD model (equation (5)). The regressions are based on data from the business censuses 1995-2008. The estimation sample comprises of all establishments existing throughout 1995-2008. The main coefficient of interest is the interaction between firms within 15 minutes commuting time d_i to the border and the free movement period, $I(t \ge 2005) \cdot I(d_i \le 15)$. The dependent variable in panel A is full-time equivalent (FTE) employment of foreign workers relative to total full-time equivalent employment in 1998. The dependent variable in panel B is the number of apprentices relative to total full-time equivalent employment in 1998. The dependent variable in panel C is a dummy variable equal to 1 if an establishment trains apprentices in a given census year. Regressions in panels A and B (except those in column 6) are weighted using establishments' average employment pre-1999 as weight. In columns 1–3, we control for (NACE rev. 1.1 two-digit) industry-period fixed effects (FE), NUTS-II-period FE, and canton-period FE, respectively. The regressions in column 4 are restricted to two-digit industries that are unaffected by the other bilateral agreements according to the classification by Buehler et al. (2011). Column 5 presents the results using the sample of establishments used in the analyses of Beerli et al. (2021) (BRSP). Column 6 does not weight observations in panels A and B by establishment size. Column 7 does not winsorize the outcomes in panels A and B at 99.99%. Standard errors are clustered by commuting zone. ***, **, *, denote statistical significance at the 1%, 5%, and 10% level, respectively.

framework can rationalize this finding if the new foreign workers caused a scale effect that offset the incentives to train less through the cost effect. This section provides evidence that both mechanisms played a role in our context.

5.5.1 The relevance of the scale effect

We start with the scale effect that arises if skilled foreign workers and native workers are imperfect substitutes. The scale effect implies that heavily exposed establishments should grow more than comparable non-treated establishments.

We explore this mechanism in Table 4, extending similar analyses by Beerli et al. (2021) to the context and the sample relevant in this paper. The table contains four panels. The first two panels analyze the effects on employment of all foreign workers (cross-border workers plus regular immigrants, panel A) and for cross-border workers only (panel B). Information on the employment of crossborder workers is only available in the censuses of 1995, 2005, and 2008. The DiD estimates thus reflect the regional differences in the growth of cross-border employment in the 2005–2008 period relative to 1995. Panel C studies the effects on the number of apprentices. Finally, panel D provides estimates of the effect of the free movement policy on establishment size—the scale effect. We present these estimates separately by broad economic sector since we know from Figure 6 that there is evidence for a displacement of apprentices in the construction sector—a finding that is confirmed in panel C of Table 4. We also observed a non-significant but economically meaningful negative effect in the public services. Thus, we expect that the scale effect may be most visible in the manufacturing and private services sectors and more muted in the construction and public services sectors.

The table confirms this prediction. We find statistically significant evidence of a quantitatively meaningful scale effect of the free movement policy in manufacturing (column 1, panel D). The policy effect on establishment size is also economically sizeable but marginally insignificant in the private service sector (column 3, panel D). The impacts are smaller and statistically insignificant in the two other sectors. The point estimate is even negative in the construction sector, and the confidence intervals rule out large positive effects. Interestingly, panel A, column 2, suggests that highly treated establishments in the construction sector do not hire more foreign workers than control establishments. They do, however, hire more cross-border workers (column 2, panel B). These two regressions thus suggest that cross-border

workers substitute apprentices in this sector (column 2, panel C) as well as regular immigrants.

Overall, we find evidence in line with our theoretical model that the scale effect offset the negative pressure on apprenticeship provision in manufacturing and, possibly, the private service sector. There is evidence for a quantitatively meaningful displacement of apprenticeships in the construction sector, where the scale effect was absent.

5.5.2 The relevance of the cost effect

We now analyze the relevance of the cost effect, which is based on the notion that a larger supply of skilled foreign workers reduced firms' problems to find suitable skilled workers. By lowering the financial consequences to hire skilled workers externally, a larger supply of skilled workers reduces the future savings in terms of hiring costs associated with training. We study the importance of this mechanism by exploiting the unique qualitative questions on firms' training motives in the costbenefit surveys (see section 4.1). Appendix Table C.3 provides the list of training motives covered in the surveys.

Our first piece of evidence consistent with the cost effect is descriptive: one of the most important motives that Swiss firms train apprentices are to attract skilled workers because it is hard to find qualified personnel on the external labor market. Other relevant motives consistent with the cost effect are to train junior workers into skilled workers whose skills match the firms' needs exactly and to save on the cost of hiring skilled workers on the external labor market (see Table 1).

A second, descriptive piece of evidence comes from the observed changes in the importance of firms' training motives between 2000 and 2009. Appendix figure C.2 shows that attracting skilled workers, qualifying junior workers into skilled workers, and securing a talent pipeline in the sector/region became less important training motives between 2000 and 2009. Consistent with an effect of the free movement policy on these motives, the reductions are most pronounced for firms situated near the border. Similarly, the training motives to save hiring costs for external workers, to avoid the risk of a wrong hiring decisions, and to avoid fluctuation by hiring specialists whose skills match the needs very closely became less important for firms farther away.

Table 5 formally analyzes whether the greater availability of skilled workers

	(1)	(2)	(3)	(4)
	FE	FE	FE	FE
	Manufac-	Construction	Private	Public
VARIABLES	turing	sector	services	services
A. Foreign worker	s / total F	TE employme	ent in 199	8
$I(t = 2001) \cdot I(d_i \le 15)$	0.032*	0.000	0.000	0.015
	(0.017)	(0.018)	(0.013)	(0.010)
$I(t \ge 2005) \cdot I(d_i \le 15)$	0.088***	-0.001	0.023***	0.040***
	(0.028)	(0.021)	(0.008)	(0.011)
B. Cross-border w	orkers / to	otal employme	nt in 199	5
$I(t \ge 2005) \cdot I(d_i \le 15)$	0.100^{***}	0.038*	0.055***	0.089**
	(0.022)	(0.019)	(0.013)	(0.040)
C. Apprentices	/ total FT	E employment	in 1998	
$I(t = 2001) \cdot I(d_i \le 15)$	0.002	-0.006**	-0.000	-0.001
	(0.003)	(0.003)	(0.002)	(0.006)
$I(t \ge 2005) \cdot I(d_i \le 15)$	-0.000	-0.009**	-0.002	-0.007
	(0.004)	(0.004)	(0.002)	(0.006)
D. 1	log FTE er	nployment		
$I(t = 2001) \cdot I(d_i \le 15)$	0.051*	0.001	0.023	0.002
, , , , ,	(0.028)	(0.025)	(0.016)	(0.012)
$I(t \ge 2005) \cdot I(d_i \le 15)$	0.082**	-0.043	0.042	0.017
<pre></pre>	(0.038)	(0.026)	(0.029)	(0.015)
	× ,	× /	· · · ·	· /
Control group	Both	Both	Both	Both
Balanced sample	Yes	Yes	Yes	Yes
Establishment FE	Yes	Yes	Yes	Yes
Period FE	Yes	Yes	Yes	Yes

Table 4: Heterogeneity by broad economic sector

Notes: This table shows the effect of the free movement policy on foreign employment, employment of cross-border workers, apprenticeships, and establishment size of highly treated establishments using our main DiD model (equation (5)). The regressions are based on data from the business censuses 1995–2008 and estimated separately by sector. The estimation sample comprises of all establishments existing throughout 1995–2008. The main coefficient of interest is the interaction between firms within 15 minutes commuting time d_i to the border and the free movement period, $I(t \ge 2004) * I(d_i \le 15)$. The effects on slightly treated firms $(I(15 < d_i \leq 30))$ are estimated but omitted here for brevity. The dependent variable in panel A is full-time equivalent (FTE) employment of foreign workers relative to total full-time equivalent employment in 1998. The dependent variable in panel B is the number of cross-border workers relative to total employment in 1998. This information is only available in the censuses 1995, 2005, and 2008, which explains why we do not estimate the effects in the transition phase (t = 2001). The dependent variable in panel C is the number of apprentices relative to total full-time equivalent employment in 1998. The dependent variable in panel D is establishments' log full-time equivalent employment. Regressions are weighted using establishments' average employment pre-1999 as weight. Standard errors are clustered by commuting zone. ***, **, *, denote statistical significance at the 1%, 5%, and 10% level, respectively.

affected firms' training motives by making it easier for firms to find skilled workers in the external labor market. Since the motives were levied on a 5-point Likert scale, we estimate equation (5) using ordered probit models.²⁵ These regression do not contain firm fixed effects because the surveys do not track firms over time. Instead, we absorb firm size fixed effects²⁶, industry fixed effects (19 industries²⁷), and fixed effects per training occupation. We estimate heteroscedasticity-robust standard errors²⁸ and weight observations using firms' sampling weights.

The estimates of the interaction term $I(d_i \leq 15) * I(t = 2009)$ provide strong evidence that the free movement policy influenced firms' training motives in a way that is consistent with the cost effect. The policy had a statistically significant negative effect on the following five training motives: it reduced highly treated firms' motives to train apprentices to attract skilled workers because it is hard to find qualified workers externally (column 1), to save hiring costs for external workers (column 2), to avoid the risk of wrong hiring decisions (column 3), to avoid high personnel fluctuation (column 7), and to qualify junior workers into skilled workers (column 8).

We can gain an intuition about the economic size of these effects if we compute predictive margins at a specific value of the 5-point survey questions. The policy, for example, reduced the probability that highly treated firms think that training to attract skilled workers is an important training motive by about 1.2 percentage points. The probability that it is a very important motive decreased by 8.3 percentage points. Similarly, the policy increased the probability that saving hiring costs

 $^{^{25}}$ Appendix table C.7 shows that the results are similar if we use a linear probability model where the outcome is 1 if the firms consider the respective motive as important or very important and zero otherwise.

 $^{^{26}}$ We add separate dummies for four broad firm size categories: 0–9 workers, 10–49 workers, 50–99 workers, and 100+ workers.

²⁷The 19 industries are construction, food product and beverage manufacturing, textile and apparel manufacturing, wood and paper product manufacturing, chemical manufacturing, metal products manufacturing, machinery and equipment manufacturing, electrical equipment manufacturing, other manufacturing, trade and repair, food and beverage service activities, transport and telecommunication, financial services and insurance, real estate, IT, education, human health and social work, public administration, and other services.

 $^{^{28}}$ We do not estimate cluster-robust standard errors at the level of labor market (NUTS-III) regions because our cost-benefit data does not contain detailed information on the location of firms due to privacy concerns. While we know the exact distance to the border for each firm—this information was computed by the data provider using our crosswalk—, the highest-resolution geographical information that we have is the *canton* in which the firm is located. Table C.8 shows ordered probit estimates with standard errors clustered at the cantonal level. The estimated standard errors are *smaller* in this case. However, these standard errors may be biased because of the small number of cantons (26).

for external workers is not an important training motive by 7.3 percentage points. Interestingly, some of these effects manifest themselves already in the 2004 survey, conducted in autumn and thus only a few months after treated firms gained unrestricted access to cross-border workers. As expected, we observe similarly signed but generally smaller effects for slightly treated firms. We find no statistically significant effect of the policy on the motives to replace unskilled workers, save on adjustment costs, and secure a talent pipeline in the sector/region.

Overall, we interpret these findings as supportive for our view that the free movement policy caused a cost effect that reduced treated firms' incentives to train apprentices. Therefore, absent the scale effect demonstrated in the previous subsection, the opening of the border could have had detrimental effects on firms' provision of apprenticeships.

	(1) Oprobit	(2) Oprobit	(3) Oprobit	(4) Oprobit	(5) Oprobit	(6) Oprobit	(7) Oprobit	(8) Oprobit	(9) Oprobit
VARIABLES	Attract skilled	Hiring cost	Risk of wrong	Replace unskilled	Adjust- ment	$\operatorname{Hire}_{\operatorname{the}}$	Avoid fluc-	Qualify junior	Secure skilled
	workers		decision	workers	cost	best	tuation	staff	workers
$I(d_{ m i} \leq 15)$	0.044	0.218^{**}	0.280^{***}	0.161^{*}	0.200^{**}	0.264^{***}	0.271^{***}	0.182^{**}	-0.113
$I(d. < 15) \cdot I(f - 2004)$	(0.090)	(0.092)	(0.100)	(0.089)	(0.095) -0 169	(0.089)	(0.093)	(0.092)-0.931*	(0.090)
	(0.139)	(0.136)	(0.138)	(0.129)	(0.135)	(0.133)	(0.135)	(0.133)	(0.134)
$I(d_{ m i} \leq 15) \cdot I(t=2009)$	-0.262^{**}	-0.286^{**}	-0.275^{**}	-0.202	-0.115	-0.117	-0.205^{*}	-0.262^{**}	0.086
	(0.130)	(0.118)	(0.115)	(0.125)	(0.126)	(0.123)	(0.123)	(0.127)	(0.147)
$I(15 < d_{ m i} < 30)$	0.145* (0.075)	0.071 (0.078)	0.078	0.077 (0.086)	0.008	0.029	0.109 (0.079)	0.119 (0.080)	-0.089
$I(15 < d_{ m i} \le 30) \cdot I(t = 2004)$	-0.373^{***}	0.088	-0.035	-0.057	0.075	-0.039	-0.122	-0.250^{**}	-0.072
	(0.110)	(0.112)	(0.116)	(0.116)	(0.111)	(0.113)	(0.112)	(0.114)	(0.116)
$I(15 < d_{ m i} \le 30) \cdot I(t = 2009)$	-0.228**	-0.077	-0.063	-0.034	0.043	-0.031	-0.179^{*}	-0.279^{**}	-0.046
	(0.116)	(0.098)	(0.100)	(0.112)	(0.105)	(0.113)	(0.102)	(0.117)	(0.119)
Observations	9068	9068	9068	9068	9068	9068	9068	9068	9068
Year FE	$\mathbf{Y}_{\mathbf{es}}$	$\mathbf{Y}_{\mathbf{es}}$	$\mathbf{Y}_{\mathbf{es}}$	$\mathbf{Y}_{\mathbf{es}}$	$\mathbf{Y}_{\mathbf{es}}$	$\mathbf{Y}_{\mathbf{es}}$	$\mathbf{Y}_{\mathbf{es}}$	$\mathbf{Y}_{\mathbf{es}}$	$\mathbf{Y}_{\mathbf{es}}$
Occupation FE	\mathbf{Yes}	$\mathbf{Y}_{\mathbf{es}}$	$\mathbf{Y}_{\mathbf{es}}$	\mathbf{Yes}	\mathbf{Yes}	${ m Yes}$	$\mathbf{Y}_{\mathbf{es}}$	$\mathbf{Y}_{\mathbf{es}}$	\mathbf{Yes}
Industry FE	\mathbf{Yes}	Yes	Yes	\mathbf{Yes}	\mathbf{Yes}	\mathbf{Yes}	$\mathbf{Y}_{\mathbf{es}}$	\mathbf{Yes}	\mathbf{Yes}
Firm Size FE	\mathbf{Yes}	Yes	Yes	\mathbf{Yes}	\mathbf{Yes}	\mathbf{Yes}	$\mathbf{Y}_{\mathbf{es}}$	\mathbf{Yes}	Yes
<i>Notes:</i> The table presents the estimatic based on ordered probit regressions of a	on results of the aggregate cross-	e free moveme sections for fi	ent policy on rms in 2000,	the training 1 2004 and 2009	notives of fir). The depen	ms in the cos dent variable	t-benefit data. s "attract skille	All DiD esti ed workers", "	mations are hiring cost",
"risk of wrong decision", "replace unskill from qualitative survey quastions on th	led workers", ad de importance o	ljustment cos of these traini	ts, choose the	e best, avoid f 5-noint Likert	luctuation, q scale when	ualify junior	staff, and "secu nortant and 5	ure skilled wc is verv imno	rkers" come rtant) The
treated groups are the firms located up	to 15 minutes	from the clos	lest border w	ithin the bord	er region (I)	$d_i \leq 15)$ and	d the firms loc	ated 15 minu	tes to up to
30 minutes from the closest border with	in the border r	egion $(I(15 <$	$d_i \leq 30)$. T	The control gro	oup consists o	of firms locate	ed more than 5	30 minutes av	ay from the
closest border within the border region	and outside of i	t. All regress	ions control f	or firm size, ir	idustry, and	occupation fix	ted effects (see	text for deta	ls). ***, **,
* denote statistical significance at the 1	%, 5%, and 10%	6 level, respect	ctively. All es	timations use	sample weig	nts and robus	t standard erre	ors.	

•	motives
•	training
c	on hrms
;	policy
-	movement
J	Iree
L ' J	of the
۔ ۲	LITECT
ו - -	Table 5:

6 Conclusion

This study analyzes the effects of opening the Swiss labor market to cross-border workers on the employment of (skilled) foreign workers, the firms' provision of apprenticeships to unskilled native workers, and the training motives of firms. We exploit the step-wise implementation of the agreement on the free movement of persons, which affected firms near the Swiss border more than firms farther away.

We show that the free movement policy increased the number of (skilled) foreign workers in the firms situated near the border that provide apprenticeships. This increase in foreign employment did not displace apprentices on aggregate. While the point estimate of our preferred specification suggests that ten additional crossborder workers replace 0.9 apprentice positions, the standard errors do not rule out a zero effect. Consistent with our theoretical framework, the policy reduced treated firms' incentives to train apprentices to save on hiring costs for external skilled workers. A scale effect counteracted the lower incentive to train apprentices arising from this cost effect: the free movement policy had a positive impact on establishment size in manufacturing and possibly in the private service sector. This effect on firm size, possibly the result of imperfect substitution between skilled immigrant and skilled native workers, allowed firms to hire more workers overall. We find evidence for a displacement of apprenticeship positions in the construction sector where we do not find evidence for a scale effect.

Our findings might suffer from two limitations. First, our estimates do not capture possible general equilibrium effects. The far-reaching reform might have affected all firms in Switzerland to some extent. Such effects are absorbed in our specification, which focuses on the differences in the policy's effects between firms that were more and less exposed. Second, our data is limited to the period after 1995. Hence, we cannot show pre-trends for an extended pre-treatment period. While the number of foreigners employed at highly treated firms exhibits a clear break when the policy takes place, the evidence is less clear for the number of apprentices. Here, we observe a gradual excess decline in highly treated firms. The gradual trend could be caused by the policy but also by other factors, which cautions us from drawing firm conclusions from the negative point estimate on the training of apprentices.

References

- Acemoglu, Daron and Joern-Steffen Pischke, "Why Do Firms Train? Theory and Evidence," The Quarterly Journal of Economics, 1998, 113 (1), 79–119.
- Aepli, Manuel and Andreas Kuhn, "Open Labor Markets and Firms 'Substitution between Training Apprentices and Hiring Workers," *Labour Economics*, June 2021, 70, 101979.
- Ariu, Andrea, "Foreign Workers, Product Quality, and Trade: Evidence from a Natural Experiment," 2020.
- Bächli, Mirjam and Teodora Tsankova, "Free Movement of Workers and Native Demand for Tertiary Education," 2020.
- Baker, Meredith and Mark Wooden, "Immigration and Its Impact on the Incidence of Training in Australia," *Australian Economic Review*, 1992, 25 (2), 39–53.
- Bassanini, Andrea, Alison Booth, Giorgio Brunello, Maria De Paola, and Edwin Leuven, "Workplace Training in Europe," in "Education and Training in Europe," Oxford University Press, 2007.
- **BBT**, "Berufsbildungsreform: häufig gestellte Fragen (FAQ)," Technical Report, BBT, Bern 2003.
- Beerli, Andreas, Jan Ruffner, Michael Siegenthaler, and Giovanni Peri, "The Abolition of Immigration Restrictions and the Performance of Firms and Workers: Evidence from Switzerland," *American Economic Review*, 2021, 111 (3), 976–1012.
- Bellmann, Lutz, Philipp Grunau, Klaus Troltsch, and Günter Walden, "Make or buy: train in-company or recruit from the labour market?," *Empirical Research in Vocational Education and Training*, 2014, 6 (1), 1–18.
- Blatter, Marc, Samuel Muehlemann, and Samuel Schenker, "The costs of hiring skilled workers," *European Economic Review*, 2012, 56 (1), 20–35.
- _ , _ , _ , **and Stefan C. Wolter**, "Hiring costs for skilled workers and the supply of firm-provided training," Oxford Economic Papers, 2016, 68 (1), 238–257.
- Blocher, Christoph, "Die Wirtschaft braucht keine Masseneinwanderung," 2011.
- Borjas, George J., "The labor demand curve is downward sloping: Reexamining the impact of immigration on the labor market," *The Quarterly Journal of Economics*, 2003, 118 (4), 1335–1374.
- Brunello, Giorgio, Elisabetta Lodigiani, and Lorenzo Rocco, "Does low skilled immigration increase the education of natives? Evidence from Italian provinces," *Labour Economics*, 2020, 63, 101794.
- Buehler, Stefan, Marco Helm, and Michael Lechner, Trade Liberalization and Growth: Plant-Level Evidence from Switzerland, Vol. August 2011.

- Campo, Francesco, Giuseppe Forte, and Jonathan Portes, "The Impact of Migration on Productivity and Native-Born Workers' Training," *IZA Discussion Paper* Series, 2018, 11833.
- Cappelli, Peter, "Why do employers pay for college?," Journal of Econometrics, 2004, 121 (1-2), 213–241.
- Card, David, "Immigrant inflows, native outflows, and the local labor market impacts of higher immigration," *Journal of Labor Economics*, 2001, 19 (1), 22–64.
- Chassamboulli, Andri and Giovanni Peri, "The economic effect of immigration policies: analyzing and simulating the U.S. case," *Journal of Economic Dynamics and Control*, 2020, 114, 103898.
- Cristelli, Gabriele and Francesco Lissoni, "Free movement of inventors: open-border policy and innovation in Switzerland," Available at SSRN, 2020.
- Dionisius, Regina, Samuel Muehlemann, Harald Pfeifer, Günter Walden, Felix Wenzelmann, and Stefan C Wolter, "Costs and Benefits of Apprenticeship Training. A Comparison of Germany and Switzerland," Applied Economics Quarterly, 2009, 55 (1), 7–37.
- Dustmann, Christian, Uta Schoenberg, and Jan Stuhler, "Labor supply shocks, native wages, and the adjustment of local employment," *The Quarterly Journal of Economics*, 2017, pp. 435–483.
- Gerfin, Michael and Boris Kaiser, "The effects of immigration on wages: An application of the structural skill-cell approach," Swiss Journal of Economics and Statistics, 2010, 146 (4), 709–739.
- Hermann, Rudolf, "Mit seiner Asylpolitik schreckt Dänemark auch Ausländer ab, die es brauchen würde," 2019.
- Jaeger, David, Joakim Ruist, and Jan Stuhler, "Shift-Share Instruments and the Impact of Immigration," *NBER Working Paper Series*, 2018.
- Leuven, Edwin, "The economics of private sector training: A survey of the literature," *Journal of Economic Surveys*, 2005, 19 (1), 91–111.
- Lindley, Robert M., "The Demand for Apprentice Recruits by the Engineering Industry, 1951-71," Scottish Journal of Political Economy, 1975, 22 (1), 1–24.
- Mansour, Hani, "The effects of labor supply shocks on labor market outcomes: Evidence from the Israeli-Palestinian conflict," *Labour Economics*, 2010, *17* (6), 930–939.
- Merrilees, William J., "Alternative models of apprentice recruitment: With special reference to the British engineering industry," *Applied Economics*, 1983, 15 (1), 1–21.
- Mohrenweiser, Jens and Thomas Zwick, "Why do firms train apprentices? The net cost puzzle reconsidered," *Labour Economics*, 2009, *16*, 631–637.

- _ , _ , and Uschi Backes-Gellner, "Poaching and Firm-Sponsored Training," British Journal of Industrial Relations, mar 2019, 57 (1), 143–181.
- Mountford, Andrew and Jonathan Wadsworth, "Trainspotting : "Good Jobs", Training and Skilled Immigration," *IZA Discussion Paper Series*, 2019, *12409*.
- Muehlemann, Samuel, "Training participation of internationalized firms: Establishment-level evidence for Switzerland," *Empirical Research in Vocational Education and Training*, 2014, 6 (5), 1–11.
- and Mirjam Strupler Leiser, "Hiring costs and labor market tightness," Labour Economics, 2018, 52, 122–131.
- and Stefan C. Wolter, "Regional Effects on Employer-Provided Training: Evidence from Apprenticeship Training in Switzerland," ZAF, 2007, 2&3, 135–147.
- and _ , "Firm-sponsored training and poaching externalities in regional labor markets," Regional Science and Urban Economics, 2011, 41, 560–570.
- _ , _ , Marc Fuhrer, and Adrian Wueest, Lehrlingsausbildung-ökonomisch betrachtet. Ergebnisse der zweiten Kosten-Nutzen-Studie, Chur, Zürich: Rueegger Verlag, 2007.
- Mueller, Barbara and Jurg Schweri, "How specific is apprenticeship training? Evidence from inter-firm and occupational mobility after graduation," Oxford Economic Papers, 2015, 67 (4), 1057–1077.
- Naguib, Costanza, "Estimating the heterogeneous impact of the free movement of persons on relative wage mobility," Available at SSRN 3344222, 2019.
- **Oatey, Michael**, "The economics of training with respect to the firm," *British Journal* of Industrial Relations1, 1970, 8 (1), 1–21.
- **OECD**, OECD Handbook for internationally comparative education statistics. Concepts, standards, definitions and classifications., Paris: OECD Publishing, 2004.
- Ottaviano, Gianmarco I.P. and Giovanni Peri, "Rethinking the effect of immigration on wages," Journal of the European Economic Association, 2012, 10 (1), 152–197.
- Schweri, Jürg, Samuel Mühlemann, Yasmina Pescio, Belinda Walther, Stefan C. Wolter, and Lukas Zürcher, Kosten und Nutzen der Lehrlingsausbildung aus der Sicht Schweizer Betriebe, 1 ed., Zürich/Chur: Verlag Rüegger, 2003.
- SERI, "Vocational and professional education and training in Switzerland: Facts and figures 2020," Technical Report, State Secretariat for Education, Research and Innovation (SERI), Bern 2020.
- Shintoyo, Naoki, "Creation of jobs and firm-sponsored training in a matching model of unemployment," Journal of Economics/ Zeitschrift fur Nationalokonomie, 2008, 93 (2), 145–176.

- __, "Worker-financed versus Firm-sponsored Training: How are Skilled Workers Supplied to the Economy?," *Economica*, 2010, 77, 110–127.
- Stevens, Margaret, "An Investment Model for the Supply of Training by Employers," *The Economic Journal*, 1994, 104 (424), 556–570.
- Strupler, Mirjam and Stefan C. Wolter, Die duale Lehre: eine Erfolgsgeschichte auch für die Betriebe, 1 ed., Glarus/Chur: Rüegger Verlag, 2012.
- Winkelmann, Rainer, "Employment prospects and skill acquisition of apprenticeshiptrained workers in Germany," *Industrial and Labor Relations Review*, 1996, 49 (4), 658–672.
- Wolter, Stefan C. and Paul Ryan, "Apprenticeship," in Eric A Hanushek, Stephen Machin, and Ludger Woessmann, eds., *Handbook of the Economics of Education*, Vol. 3, Amsterdam: Elsevier B.V., 2011, book section 11, pp. 521–576.
- _ , Samuel Muehlemann, and Juerg Schweri, "Why Some Firms Train Apprentices and Many Others Do Not," German Economic Review, 2006, 7 (3), 249–264.

Appendix

A Solving the model

In this section, we derive the first-order condition of the firms' optimization problem sketched in section 2 of the main paper. Recall that the optimization problem reads as follows:

$$\max_{T_0, N_0, I_0} A_0 (\alpha N_0^{\rho} + (1-\alpha)I_0^{\rho})^{\frac{1}{\rho}} - w_0^N N_0 - w_0^I I_0 - \frac{1}{2}c_0 T_0^2 - \frac{1}{2}b_0 [L^s](N_0 + I_0)^2 + A_1 (\alpha N_1^{\rho} + (1-\alpha)I_1^{\rho})^{\frac{1}{\rho}} - w_1^N N_1 - w_1^I I_1 - \frac{1}{2}b_1 [L^s] X_1^2 \quad (A.1)$$

The first-order condition for skilled native and immigrant workers are:

$$N_0 : A_0 \alpha N_0^{\rho-1} (\alpha N_0^{\rho} + (1-\alpha) I_0^{\rho})^{\frac{1-\rho}{\rho}} - w_0^N - b_0 [L^s] (N_0 + I_0) + (1-\delta) b_1 [L^s] ((N_1 + I_1) - (1-\delta) (N_0 + I_0) - (1-\gamma) T_0) = 0 \quad (A.2)$$

$$I_0: A_0(1-\alpha)I_0^{\rho-1}(\alpha N_0^{\rho} + (1-\alpha)I_0^{\rho})^{\frac{1-\rho}{\rho}} - w_0^I - b_0[L^s](N_0 + I_0) + (1-\delta)b_1[L^s]((N_1 + I_1) - (1-\delta)(N_0 + I_0) - (1-\gamma)T_0) = 0 \quad (A.3)$$

The first-order condition for training unskilled apprentices reads as follows:

$$T_0: -c_0 T_0 + (1-\gamma)b_1[L^s]((N_1 + I_1) - (1-\delta)(N_0 + I_0) - (1-\gamma)T_0) = 0$$
 (A.4)

To derive an expression for firms' employment of trainees, we follow Stevens' (1994) and define H_0 as a function of exogenous variables reflecting the ratio between hiring costs of skilled workers and training costs unskilled workers, i.e. $H_0 \equiv c_0/(b_1[L^s](1-\gamma)^2)$ with $H_0 \geq 0$; $\frac{\partial H_0}{\partial b_1} \leq 0$; $\frac{\partial H_0}{\partial c_1} \geq 0$ because $c_t \geq 0$. Combining the three first-order conditions—the two for skilled workers (eq. 4 and eq. 5) and the first-order condition for training (eq. 6)—, and rearranging yields the following decision rule for the training of unskilled workers:

$$T_0 = \frac{1}{H_0 + 1} ((N_1 + I_1) - (1 - \delta)I_0(G_0 + 1))$$
(A.5)

where we express the first-period use of skilled labor $N_0 + I_0$ as a function of I_0 and G_0 which relates the first-period wages of skilled native to those of skilled immigrant workers, that is, $G_0 \equiv \left(\frac{1-\alpha}{\alpha} \frac{w_0^N}{w_0^I}\right)^{-\sigma}$.

Equation A.5 shows that firms train unskilled workers to cover a fraction $1/(1 + H_0)$ of the additionally needed skilled workers in the future. In general, the number of unskilled workers that firms train depends on current and future employment of native and immigrant workers and on the relative wages between skilled immigrant and skilled native workers (to an extent that is governed by the elasticity of substitution between the two skilled types). It also depends on the training costs relative to the hiring costs: the lower the training costs relative to the hiring costs, the larger the number of trainees and vice versa. Firms do not train when the hiring costs b_1 approach zero. Conversely, when training costs c_0 are zero, firms train all additional skilled workers needed for the future.

To derive the decision rule for skilled native and skilled immigrant workers, one can apply the standard CES solution and express the log relative demand for native and skilled workers as a function of the log ratio wages of skilled native to skilled immigrant workers:

$$\log \frac{N_0}{I_0} = \sigma * \left(\log \frac{\alpha}{1-\alpha} - \log \frac{w_0^N}{w_0^I}\right). \tag{A.6}$$

The employment decision of skilled workers, as in Stevens (1994), does not depend on the number of trained unskilled workers.

B Results with the matched control group

This section presents estimates of the effect of the free movement policy on foreign employment and the provision of apprenticeships if we use control establishments matched to highly treated units based on Mahalanobis distance matching. Since the firms that we compare are thus similar along a set of observed pre-treatment variables such as industry affiliation and international exposure, these results are more robust to unobserved shocks that affect firms along the matched attributes or attributes correlated with the matched attributes.

To construct the matched sample, we only focus on the balanced sample of establishments existing in all years 1995–2008. We then match control establishments—establishments located more than 30 minutes away from the border in either region—to highly treated establishments—establishments located within 15 minutes to the border in the border region—using Mahalanobis distance matching. We only consider control units that completely agree with highly treated units in terms of indicators whether a firm trained apprentices, employed foreign workers, of two-digit industry affiliation (NACE rev. 1.1), of establishment size (in 4 groups), all measured in 1998, and whether a firm existed in 1991. These restrictions drop 133 highly treated establishments without corresponding control unit. We additionally match on the following pre-treatment covariates: (continuous) establishment size (in full-time equivalents) in 1998, the foreign employment share in 1995, and indicators whether the establishment is publicly owned, a single firm, a subsidiary, a headquarter (all in 1998), an exporter, and an importer (both in 1995). The same establishment can serve as control for several highly treated units. We randomly select one control establishment if there are several control units that have the same Mahalanobis distance score.

Table B.1 table provides descriptive statistics of certain pre-treatment characteristics of the highly treated establishments and the matched control group. The characteristics are measured in 1998 unless otherwise noted. The number of observations differs between the two groups because the average control establishment is matched to 2.5 treated units. However, we account for the unequal number of observations when calculating means and standard deviations in the other rows of columns 3–4 by weighting them by the number of times that an untreated establishment is matched to a treated establishment. The table shows that the matched control group is, sometimes by construction, statistically indistinguishable from the treatment group along some important pre-treatment characteristics. As expected, firms in the high-treatment group have a higher share of cross-border workers. As a consequence, they also have a somewhat higher share of foreigners in total employment. But since we match on these firm characteristics, all observed differences are considerable smaller than in our baseline sample (see Table 1).

Table B.2 shows the results if we use the matched control group to estimate our main DiD model.²⁹ The regressions are weighted by number of times that an untreated establishment is matched to a treated establishment. We additionally multiply this weight by establishments' average employment in the estimation period, topcoded at 500 FTE workers per establishment (as in our baseline regressions). Control establishments that are never matched are dropped. Reassuringly, the table shows that the effect estimates that we get are close to our baseline estimates in column 2 of Table 2, independent of whether we control for the potential supply of apprentices in the regional labor market (column 2 and 4) or not (columns 1 and 3).

Table B.1: Pre-treatment characteristics of highly treated establishments and matched control group

	(1)	(2)	(3)	(4)	(5)	(6)
	Hi	ghly	Ma	tched	Differe	nce
	tre	ated	coi	\mathbf{ntrol}	in mea	ans
Variable	Mean	SD	Mean	SD	Difference	SE
FTE employment	11.286	(56.671)	10.454	(43.263)	0.833**	(0.413)
Existed in 1991	0.796	(0.403)	0.796	(0.403)	0.000	(0.003)
Employs for eigners $(0/1)$	0.477	(0.499)	0.477	(0.499)	-0.000	(0.004)
Share of foreign workers	0.230	(0.315)	0.201	(0.291)	0.029^{***}	(0.002)
Cross-border worker share (1995)	0.075	(0.168)	0.004	(0.038)	0.072^{***}	(0.001)
Training firm $(0/1)$	0.209	(0.406)	0.209	(0.406)	-0.000	(0.003)
Apprentice share	0.044	(0.111)	0.046	(0.112)	-0.001	(0.001)
Manufacturer $(0/1)$	0.114	(0.318)	0.114	(0.318)	0.000	(0.003)
High-tech manufacturing $(0/1)$	0.032	(0.175)	0.032	(0.175)	-0.000	(0.001)
Construction $(0/1)$	0.086	(0.280)	0.086	(0.280)	0.000	(0.002)
Publicly owned firm $(0/1)$	0.124	(0.330)	0.123	(0.329)	0.001	(0.003)
Exporter $(0/1, 1995)$	0.163	(0.369)	0.157	(0.364)	0.006^{*}	(0.003)
Importer $(0/1, 1995)$	0.258	(0.437)	0.256	(0.436)	0.002	(0.004)
Travel minutes to border	7.129	(3.544)	47.567	(13.661)	-40.437***	(0.082)
Observations	42,490		17,068		59,558	

Notes: The table shows mean and standard deviation of pre-treatment characteristics of highly treated establishments (establishments located within 15 minutes to the border in the border region, columns 1 and 2) and a matched control group of establishments located in one of the two control regions (establishments located more than 30 minutes to the border in the border region and establishments in the non-border region, columns 3 and 4). The characteristics are measured in 1998 unless otherwise noted. Columns 5 and 6 test whether the covariates are balanced in treated and control units. Details on the matching are given in the text. A given control observation may appear as a match for more than one treated observation. The average control establishment is matched to 2.5 treated units (hence the difference in the number of observations). Mean and standard deviation in columns 3–4 are weighted by the number of times that an untreated establishment is matched to a treated establishment. ***, **, *, denote statistical significance at the 1%, 5%, and 10% level, respectively.

 $^{^{29}}$ The corresponding event study is panel (b) of Figure 5 in the main text.

	(1)	(2)	(3)	(4)
	\mathbf{FE}	\mathbf{FE}	\mathbf{FE}	${ m FE}$
	Foreigners/	$\mathbf{Foreigners}/$	Apprentices/	$\operatorname{Apprentices}/$
	total FTEs	total FTEs	total FTEs	total FTEs
VARIABLES	1998	1998	1998	1998
$I(t = 2001) \cdot I(d_i \le 15)$	0.021^{***}	0.020^{**}	0.000	0.000
	(0.008)	(0.010)	(0.002)	(0.002)
$I(t \ge 2005) \cdot I(d_i \le 15)$	0.050^{***}	0.047^{***}	-0.004	-0.003
	(0.010)	(0.012)	(0.003)	(0.002)
Observations	297,790	297,790	297,790	297,790
R-squared	0.576	0.576	0.605	0.605
Control group	Matched	Matched	Matched	Matched
Firm FE	Yes	Yes	Yes	Yes
Period FE	Yes	Yes	Yes	Yes
Apprentice supply control	No	Yes	No	Yes

Table B.2: Main results using matched control group

Notes: This table shows the effect of the free movement policy on foreign employment and the provision of apprenticeships if we use control establishments matched to highly treated units based on Mahalanobis distance matching. The regressions are based on data from the business censuses (BC) 1995–2008. The dependent variable in columns 1–2 is full-time equivalent (FTE) employment of foreign workers relative to total full-time equivalent employment in 1998. The dependent variable in columns 3–4 is the number of apprentices relative to total full-time equivalent employment in 1998. The border and the free movement period, $I(t \ge 2004) * I(d_i \le 15)$. The control group is establishments located more than 30 minutes away from the border matched to highly treated units with Mahalanobis distance matching. Details on the matching are given in the associated text. The regressions are weighted by number of times that an untreated establishment is matched to a treated establishment. We then multiply this weight by establishments' average employment pre-1999. The "apprentice supply control" is the (estimated) number of graduates from lower secondary schools (11th grade) in the commuting zone (NUTS-III region). ***, **, *, denote statistical significance at the 1%, 5%, and 10% level, respectively.

C Further Tables and Figures

Figure C.1: Employment share of apprentices in 1998 and change in employment share of cross-border workers, by occupation



Notes: This figure compares the employment share of recent (native) apprentices in 1998 in an occupation with the change in the employment share of cross-border workers in total employment between 1998 and 2010. The figure is based on data from the Swiss Earnings Structure Surveys 1998 and 2010. The employment share is the fraction that the two worker categories make up in the specific occupation in the border region. We focus on workers aged 18–65 in the private sector, with non-missing information for nationality, place of work, education, wages, and full-time equivalents. Cross-border workers are identified based on their residency permit. Recent native apprenticeship graduates are Swiss workers aged 18–29 with an apprenticeship as highest degree. The same data is shown in Table C.2 in tabular form.



Figure C.2: Importance of firms' training motives, by region

Notes: This figure shows firms' motives to provide apprenticeships according to the cost benefit surveys in 2000 (blue) and 2009 (teal), as well as the change between 2000 and 2009 (green). We differentiate highly treated firms (firms within the border region within 15 minutes to the border), slightly treated firms (firms within the border region 15–30 minutes away from the border) and firms in the two control groups (firms within the border region more than 30 minutes away from the border and firms outside the border region).

Table C.1	: Characteristics	of recent	apprenticeship	graduates,	natives,	and	cross-
border wo	rkers in the bord	er region,	1998 and 2010				

		1998			2010	
Worker characteristics	Recent apprentices	Cross-border workers	Other natives	Recent apprentices	Cross-border workers	Other natives
Demographic characteristics						
Mean age	24.897	39.658	42.557	24.516	40.542	44.234
Share male	0.528	0.693	0.614	0.508	0.648	0.531
Mean tenure	3.558	9.472	10.573	2.816	7.284	9.827
Mean log hourly real wage	3.311	3.455	3.620	3.321	3.545	3.687
Share tertiary educated	0.000	0.153	0.235	0.000	0.291	0.344
Share secondary educated	1.000	0.513	0.588	1.000	0.485	0.541
Share primary educated	0.000	0.334	0.178	0.000	0.224	0.115
Occupations						
Manufacture	0.125	0.286	0.111	0.105	0.210	0.070
Construction	0.092	0.121	0.053	0.102	0.097	0.039
Machine operators	0.069	0.064	0.058	0.070	0.072	0.051
Define goal & strategy	0.002	0.010	0.039	0.002	0.017	0.034
Accounting, HR	0.047	0.026	0.059	0.036	0.029	0.057
Clerks	0.105	0.023	0.073	0.064	0.028	0.057
Other clerical occupations	0.113	0.047	0.080	0.089	0.055	0.074
Logistics, strategy department	0.010	0.020	0.026	0.014	0.026	0.022
Review, consult, certify	0.049	0.012	0.058	0.041	0.033	0.069
Retail	0.126	0.060	0.101	0.155	0.069	0.091
R&D	0.008	0.038	0.017	0.006	0.049	0.023
Analyze, program, operating	0.019	0.034	0.030	0.020	0.041	0.032
Plan, design	0.036	0.030	0.045	0.032	0.034	0.032
Transport	0.031	0.069	0.062	0.036	0.046	0.043
Security	0.003	0.001	0.004	0.024	0.005	0.021
Medical, social tasks	0.063	0.036	0.053	0.078	0.068	0.087
Manicure, cleaning	0.015	0.014	0.016	0.024	0.021	0.032
Education	0.008	0.006	0.027	0.014	0.025	0.095
Restaurants and hospitality	0.042	0.057	0.035	0.064	0.056	0.039
Culture, sport, information	0.009	0.002	0.010	0.014	0.009	0.020
Others	0.015	0.010	0.025	0.010	0.009	0.011
Industries						
Agriculture/Fishing/Mining	0.003	0.005	0.004	0.013	0.006	0.006
Manufacturing	0.215	0.461	0.268	0.167	0.361	0.172
Utilities	0.002	0.001	0.007	0.007	0.003	0.011
Construction	0.093	0.127	0.062	0.106	0.102	0.049
Wholesale/Retail/Repair	0.242	0.144	0.188	0.258	0.146	0.164
Hotel/Restaurants	0.045	0.055	0.034	0.061	0.048	0.033
Transport/Communication	0.062	0.064	0.086	0.056	0.056	0.063
Financial Intermediation	0.113	0.021	0.104	0.056	0.024	0.085
Real Estate/R&D/IT	0.114	0.056	0.112	0.114	0.117	0.123
Education	0.007	0.007	0.024	0.015	0.028	0.112
Health	0.073	0.042	0.083	0.106	0.088	0.138
Personal Services	0.031	0.016	0.028	0.042	0.024	0.045
Number of Workers	163 977	103 885	859 185	102 077	185 661	1 358 600

Notes: This table shows descriptive statistics of recent native apprenticeship graduates, native workers, cross-border workers in 1998 and their change between 1998 and 2010. The table is based on data from the Swiss Earnings Structure Surveys 1998 and 2010. We focus on workers in the border region aged 18–65 working in the private sector, with non-missing information for nationality, place of work, education, wages, and full-time equivalents. Native workers are Swiss nationals, either born in Switzerland or naturalized. Cross-border workers are identified based on their residency permit. Recent native apprenticeship graduates are Swiss workers aged 18–29 with an apprenticeship as highest degree.

	Rec	ent appren	tices	Cross	-border we	orkers
Employment share	1998	2010	$\Delta 2010 - 1998$	1998	2010	$\Delta 2010 \\ -1998$
Employment share by occupation						
Manufacture	0.064	0.073	0.009	0.137	0.183	0.046
Construction	0.089	0.110	0.020	0.108	0.129	0.020
Machine operators	0.085	0.083	-0.002	0.081	0.117	0.036
Define goal & strategy	0.005	0.005	0.000	0.025	0.056	0.031
Accounting, HR	0.076	0.052	-0.024	0.040	0.056	0.015
Clerks	0.130	0.090	-0.040	0.028	0.052	0.023
Other clerical occupations	0.113	0.085	-0.028	0.045	0.066	0.021
Logistics, strategy department	0.035	0.044	0.009	0.073	0.102	0.029
Review, consult, certify	0.080	0.045	-0.036	0.018	0.048	0.030
Retail	0.087	0.103	0.016	0.046	0.067	0.022
R&D	0.034	0.014	-0.020	0.154	0.158	0.004
Analyze, program, operating	0.052	0.041	-0.011	0.094	0.108	0.014
Plan, design	0.073	0.073	0.000	0.055	0.095	0.040
Transport	0.038	0.049	0.011	0.078	0.082	0.004
Security	0.065	0.094	0.029	0.052	0.027	-0.025
Medical, social tasks	0.102	0.061	-0.041	0.051	0.081	0.030
Manicure, cleaning	0.045	0.034	-0.011	0.040	0.040	0.000
Education	0.029	0.013	-0.016	0.019	0.029	0.010
Restaurants and hospitality	0.055	0.069	0.015	0.073	0.080	0.007
Culture, sport, information	0.072	0.049	-0.022	0.016	0.042	0.026
Others	0.045	0.048	0.002	0.032	0.063	0.031
Employment share by education						
Share tertiary educated	0.000	0.000	0.000	0.061	0.079	0.018
Share secondary educated	0.161	0.139	-0.022	0.066	0.074	0.008
Share primary educated	0.000	0.000	0.000	0.104	0.108	0.004
Employment share by industry						
Agriculture/Fishing/Mining	0.061	0.138	0.077	0.082	0.084	0.002
Manufacturing	0.055	0.058	0.002	0.115	0.167	0.052
Utilities	0.032	0.056	0.025	0.013	0.032	0.018
Construction	0.083	0.104	0.020	0.112	0.131	0.019
Wholesale/Retail/Repair	0.092	0.094	0.003	0.055	0.080	0.025
Hotel/Restaurants	0.059	0.077	0.017	0.030 0.072	0.085	0.013
Transport/Communication/Storage	0.053	0.055	0.001	0.051	0.074	0.022
Financial Intermediation	0.092	0.051	-0.040	0.017	0.032	0.015
Real Estate/R&D/IT/Business	0.086	0.058	-0.028	0.041	0.080	0.039
Education	0.028	0.000	-0.016	0.027	0.028	0.001
Health	0.026 0.075	0.012 0.054	-0.010	0.021	0.020	0.030
Personal Services	0.083	0.069	-0.021	0.046	0.005	0.000
	0.000	0.003	-0.014	0.040	0.000	0.003

Table C.2: Employment shares of recent apprenticeship graduates and cross-border workers in the border region, 1998 and 2010

Number of Workers163,977192,97729,000119,962217,64997,687Notes:This table shows the share of recent apprentices and cross-border workers in total employment in the borderregion in 1998 and 2010.It uses data from the Swiss Earnings Structure Surveys 1998 and 2010.The employment is the border region.share is the fraction that the two worker categories make up in the specific occupation/industry in the border region.We focus on workers aged 18–65 working in the private sector, with non-missing information for nationality, place of work, education, wages, and full-time equivalents.Cross-border workers are identified based on their residency permit.Recent native apprenticeship graduates are Swiss workers aged 18–29 with an apprenticeship as highest degree.

Variable name	Question
	Training firm: For your firm, how important are the fol- lowing reasons to train apprentices? Non-training firm: Assuming that your firm meets all the requirements and decided to start training apprentices, how important would the following aspects of training ap- prentices be for your firm?
Attract skilled workers	Attracting skilled workers because it is hard to find quali- fied personnel on the external labor market
Hiring cost	Saving cost to hire personnel on the external labor market
Risk of wrong decision	Avoiding the risk of wrong hiring decisions that comes with external hiring
Replace unskilled workers	Replacing unskilled and semi-skilled workers with appren- tices' work
Adjustment cost	Saving the money used for training external specialists (ad- justment cost)
Hire the best	Having the opportunity to hire the "best" young person as apprentices
Avoid fluctuation	Avoiding high turnover by hiring specialists whose skills match the firm's needs very closely
Qualify junior staff	Training junior workers into skilled workers whose skills exactly match the firm's requirements
Secure skilled workers	Securing a talent pipeline in the sector/region

Table C.3: Questions on firms' training motives in the cost-benefit data

Notes: The table presents the questions used in the cost-benefit survey to inquire the training motives of firms. Training firms got a direct question on their reason to train. Non-training firms are asked a hypothetical question if they immediately started with training. All firms rated the importance of each training motive on a 5-point Likert scale.

	(1)	(2)	(3)	(4)
	FE	\mathbf{FE}	$\rm FE$	\mathbf{FE}
	For eign workers/	Number of	Training firms/	Apprentices/
VARIABLES	FTEs 1998	training firms	establishments	FTEs 1998
$I(t = 2001) \cdot I(d_i \le 15)$	0.011^{**}	0.543	-0.001	-0.000
	(0.005)	(0.597)	(0.002)	(0.001)
$I(t = 2001) \cdot I(15 < d_i \le 30)$	0.013**	1.750^{*}	0.003	0.001
	(0.005)	(0.885)	(0.002)	(0.001)
$I(t \ge 2005) \cdot I(d_i \le 15)$	0.051^{***}	0.656	-0.006	-0.003*
	(0.017)	(0.704)	(0.005)	(0.002)
$I(t \ge 2005) \cdot I(15 < d_i \le 30)$	0.021***	2.674**	0.003	0.001
	(0.006)	(1.162)	(0.002)	(0.002)
Observations	12,934	12,934	12,934	12,934
R-squared	0.205	0.052	0.299	0.226
Number of municipalities	2,594	2,594	2,594	2,594
Control group	Both	Both	Both	Both
Municipality FE	Yes	Yes	Yes	Yes
Period FE	Yes	Yes	Yes	Yes
Weights	Employment	No	Establishments	Employment

Table C.4: Municipality-level estimates of the effect of free movement policy on immigrant employment and apprenticeship training

Notes: This table shows the effect of the free movement policy on employment of foreign workers and the provision of apprenticeships using our main DiD model (equation (5)). The regressions are based on data from the business censuses (BC) 1995-2008, aggregated to the municipality level. The dependent variable in column 1 is full-time equivalent (FTE) employment of foreign workers (cross-border workers plus foreign resident immigrant workers) relative to total full-time equivalent employment in 1998 in the municipality. The dependent variable in column 2 is the number of establishments that train apprentices. The dependent variable in column 3 is the share of establishments with apprenticeships among all establishments. The dependent variable in column 4 is the number of apprentices working in a specific municipality relative to total full-time equivalent employment in 1998. The effect of the free movement period is captured with a dummy variable equal to one in the census years after 2004, i.e., the BC 2005 and 2008. The main coefficient of interest is the interaction between firms within 15 minutes commuting time d_i to the border and the free movement period, $I(t \ge 2005) \cdot I(d_i \le 15)$. The control group is firms located in the border region with more than 30 minutes travel time to the border and firms in the non-border region. Regressions in columns 1 and 4 are weighted using average municipal employment pre-1999 as weight. The regression in column 3 is weighted using the average number of establishments in a municipality as weight. Standard errors are clustered on the level of commuting zones. ***, **, *, denote statistical significance at the 1%, 5%, and 10% level, respectively.

		All firms		Trainin	g firms
	(1)	(2)	(3)	(4)	(5)
	$\dot{\mathbf{F}}\dot{\mathbf{E}}$	$\dot{\mathbf{F}}\dot{\mathbf{E}}$	$\dot{\mathbf{F}}\dot{\mathbf{E}}$	$\widetilde{\mathbf{FE}}$	FÉ
	Training	Number of	Apprentice	Number of	Apprentice
VARIABLES	firm	apprentices	share	apprentices	share
$I(d_{\rm i} \le 15)$	-0.093***	-0.137***	-0.022*	-0.031	0.002
	(0.029)	(0.043)	(0.011)	(0.055)	(0.030)
$I(d_{\rm i} \le 15) \cdot I(t = 2004)$	0.034	0.030	0.004	-0.041	-0.000
	(0.043)	(0.062)	(0.016)	(0.078)	(0.037)
$I(d_{\rm i} \le 15) \cdot I(t = 2009)$	0.069	0.127	0.020	0.138	-0.024
	(0.057)	(0.093)	(0.016)	(0.092)	(0.033)
$I(15 < d_{\rm i} < 30)$	-0.075***	-0.096**	-0.022**	0.009	-0.010
	(0.027)	(0.040)	(0.009)	(0.048)	(0.018)
$I(15 < d_{\rm i} \le 30) \cdot I(t = 2004)$	0.020	0.038	0.013	0.061	0.032
	(0.037)	(0.055)	(0.014)	(0.067)	(0.028)
$I(15 < d_{\rm i} \le 30) \cdot I(t = 2009)$	0.056	0.072	0.030**	0.092	0.014
	(0.045)	(0.073)	(0.013)	(0.066)	(0.024)
Observations	9068	9068	9041	3837	3837
Year FE	Yes	Yes	Yes	Yes	Yes
Occupation FE	Yes	Yes	Yes	Yes	Yes
Industry FE	Yes	Yes	Yes	Yes	Yes
Firm Size FE	Yes	Yes	Yes	Yes	Yes

Table C.5: Effect of the free movement policy on firms' training behavior

Notes: The table presents the estimation results of the free movement policy on the training behavior of firms in the cost-benefit data. The first three DiD estimations are based on OLS regressions of aggregate cross-sections for all firms in 2000, 2004 and 2009. The latter two DiD estimations are based on OLS regressions of aggregate crosssections for training firms in 2000, 2004 and 2009. The dependent variable in column 1 is 0 for firms not training apprentices and 1 otherwise. The dependent variable in columns 2 and 4 is 0 for firms not training apprentices and the number of apprentices otherwise. The dependent variable in columns 3 and 5 is 0 for firms not training apprentices and else the fraction of apprentices among all workers. The treated groups are the firms located up to 15 minutes from the closest border within the border region $(I(d_i \leq 15))$ and the firms located 15 minutes to up to 30 minutes from the closest border within the border region $(I(15 < d_i \leq 30))$. The control group consists of firms located more than 30 minutes away from the closest border within the border region and outside of it. ***, **, * denote statistical significance at the 1%, 5%, and 10% level, respectively. All estimations include sample weights and robust standard errors.

		All firms		Trainin	g firms
	(1)	(2)	(3)	(4)	(5)
	Ϋ́Ε	$\widetilde{\mathbf{FE}}$	$\widetilde{\mathrm{FE}}$	$\widetilde{\mathbf{FE}}$	$\widetilde{\mathbf{FE}}$
	Training	Number of	Apprentice	Number of	Apprentice
VARIABLES	firm	apprentices	share	apprentices	share
$I(d_{\rm i} \le 15)$	-0.093*	-0.146*	-0.002	-0.088	0.056
	(0.051)	(0.075)	(0.023)	(0.106)	(0.077)
$I(d_{\rm i} \le 15) \cdot I(t = 2004)$	-0.017	-0.049	-0.016	-0.050	-0.008
	(0.073)	(0.105)	(0.032)	(0.138)	(0.092)
$I(d_{\rm i} \le 15) \cdot I(t = 2009)$	0.028	0.052	0.004	0.091	0.000
	(0.071)	(0.109)	(0.030)	(0.152)	(0.081)
$I(15 < d_{\rm i} < 30)$	-0.086*	-0.132**	-0.027^{*}	-0.044	-0.030
	(0.046)	(0.065)	(0.015)	(0.086)	(0.033)
$I(15 < d_{\rm i} \le 30) \cdot I(t = 2004)$	0.018	0.025	0.014	0.044	0.053
	(0.065)	(0.093)	(0.023)	(0.120)	(0.051)
$I(15 < d_{\rm i} \le 30) \cdot I(t = 2009)$	0.032	0.063	0.018	0.052	0.036
	(0.062)	(0.091)	(0.020)	(0.119)	(0.044)
Observations	2580	2580	2571	1148	1148
Year FE	Yes	Yes	Yes	Yes	Yes
Occupation FE	Yes	Yes	Yes	Yes	Yes
Industry FE	Yes	Yes	Yes	Yes	Yes
Firm Size FE	Yes	Yes	Yes	Yes	Yes

Table C.6: Effect of the free movement policy on firms' training behavior, excluding occupations affected by the 2004 educational reform

Notes: The table presents the estimation results of the free movement policy on the training behavior of firms in the cost-benefit data for occupations not affected by the educational reform in 2003. The first three DiD estimations are based on OLS regressions of aggregate cross-sections for all firms in 2000, 2004 and 2009. The latter two DiD estimations are based on OLS regressions of aggregate cross-sections for training firms in 2000, 2004 and 2009. The latter two DiD estimations are based on OLS regressions of aggregate cross-sections for training firms in 2000, 2004 and 2009. The latter two DiD estimations are based on OLS regressions of aggregate cross-sections for training firms in 2000, 2004 and 2009. The dependent variable in columns 1 is 0 for firms not training apprentices and 1 otherwise. The dependent variable in columns 3 and 5 is 0 for firms not training apprentices and else the fraction of apprentices among all workers. The treated groups are the firms located up to 15 minutes from the closest border within the border region ($I(d_i \leq 15)$) and the firms located 15 minutes to up to 30 minutes from the closest border within the border region ($I(15 < d_i \leq 30)$). The control group consists of firms located more than 30 minutes away from the closest border within the border region and outside of it. ***, **, * denote statistical significance at the 1%, 5%, and 10% level, respectively. All estimations include sample weights and robust standard errors.

	(1) FE	FE	(3) FE	(4)FE	(5)	(6)	(7) FE	(8) FE	E) FE
Att. ADTADTEC	tract	Hiring	Risk of	Replace	Adjust-	Hire	Avoid	Qualify	Secure
VALADLED COLLECTION VOL	unea vrkers	cost	decision	workers	cost	best	tuation	staff	workers
$I(d_{\rm i} \le 15) \tag{0.0}$.023	0.057	0.108^{***}	0.032	0.088^{**}	0.070^{*}	0.119^{***}	0.087^{**}	-0.042
(0.0	.035)	(0.036)	(0.040)	(0.036)	(0.040)	(0.040)	(0.040)	(0.038)	(0.036)
$I(d_{\rm i} \le 15) \cdot I(t = 2004)$ -0.0	0.049	-0.059	-0.145^{***}	-0.026	-0.093	-0.047	-0.161^{***}	-0.125^{**}	0.048
$I(d_{\rm i} \le 15) \cdot I(t = 2009)$ -0.11	.001) 125**	$(2000) - 0.078^{*}$	-0.138^{***}	-0.063	-0.036	-0.032	(eco.o)	(1000) -0.096*	0.005
(0.0	(059)	(0.044)	(0.048)	(0.050)	(0.063)	(0.060)	(0.065)	(0.054)	(0.054)
$I(15 < d_{\rm i} < 30) $ 0.0	.041	0.012	0.027	0.031	-0.010	-0.008	0.030	0.037	-0.025
(0.0	.030)	(0.031)	(0.034)	(0.032)	(0.034)	(0.035)	(0.036)	(0.035)	(0.032)
$I(15 < d_{\rm i} \le 30) \cdot I(t = 2004) -0.13$	39^{***}	0.008	0.013	-0.011	0.027	-0.026	-0.081	-0.087*	-0.050
(0.0	.046)	(0.044)	(0.049)	(0.046)	(0.048)	(0.051)	(0.051)	(0.050)	(0.047)
$I(15 < d_{\rm i} \le 30) \cdot I(t = 2009) -0.03$	094^{**}	-0.022	-0.029	-0.050	0.048	-0.026	-0.071	-0.105^{**}	-0.043
(0.0	.045)	(0.039)	(0.043)	(0.044)	(0.048)	(0.048)	(0.052)	(0.046)	(0.045)
Observations 90	068	9068	9068	9068	9068	9068	9068	9068	9068
Vear FF	Ves	$V_{ m oc}$	Vac	$V_{ m ec}$	Vec	Vas	Vas	$V_{ m ec}$	Vas
Occupation FE	Ves	Ves	Yes	Ves	Ves	Ves	Ves	Ves	Ves
Industry FF.	Yes	Yes	γ_{es}	γ_{es}	Yes	Yes	Yes	Yes	Yes
Firm Size FE	Yes	Yes	${ m Yes}$	Yes	Yes	Yes	Yes	${ m Yes}$	Yes

	(1) Oprobit	(2) Onrohit	(3) Onrohit	(4) Onrohit	(5) Onrohit	(6) Onrohit	(7) Onrohit	(8) Onrohit	(9) Onrohit
VARIABLES	Attract skilled	Hiring	Risk of wrong	Replace unskilled	Adjust- ment	Hire the	Avoid fluc-	Qualify junior	Secure skilled
	workers		decision	workers	cost	best	tuation	staff	workers
$I(d_{ m i} \leq 15)$	0.044	0.218^{**}	0.280^{*}	0.161^{*}	0.200^{*}	0.264^{***}	0.271^{**}	0.182	-0.113
$I(d < 15) \cdot I(t - 2004)$	(0.082)	(0.107)	(0.144)	(0.095)	(0.118)	(0.099)	(0.105)	(0.116)	(0.078)
$(\pm 0.07 - 3) = (0.1 \le 10) =$	(0.106)	(0.155)	(0.147)	(0.133)	(0.149)	(0.090)	(0.149)	(0.136)	(0.129)
$I(d_{ m i} \leq 15) \cdot I(t=2009)$	-0.262^{**}	-0.286^{**}	-0.275^{*}	-0.202^{**}	-0.115	-0.117	-0.205^{*}	-0.262^{*}	0.086
1/15 > A > 20	(0.129)	(0.123)	(0.162)	(0.091)	(0.139)	(0.136)	(0.118)	(0.135)	(0.118)
$1(10 < a_{\rm i} < 30)$	(0.064)	(0.059)	0.073)	(0.106)	(0.063)	(0.080)	(0.071)	(0.062)	-0.069)
$I(15 < d_{ m i} \le 30) \cdot I(t = 2004)$	-0.374^{***}	0.088	-0.035	-0.058	0.075	-0.039	-0.122	-0.250^{***}	-0.072
	(0.079)	(0.057)	(0.059)	(0.119)	(0.070)	(0.067)	(0.087)	(0.074)	(860.0)
$I(15 < d_{\rm i} \le 30) \cdot I(t = 2009)$	-0.228**	-0.077	-0.063	-0.034	0.043	-0.031	-0.179	-0.279*	-0.046
	(0.112)	(0.089)	(0.092)	(0.140)	(0.070)	(0.119)	(0.113)	(0.156)	(0.135)
Observations	9048	9048	9048	9048	9048	9048	9048	9048	9048
Year FE	Yes	\mathbf{Yes}	$\mathbf{Y}_{\mathbf{es}}$	$\mathbf{Y}_{\mathbf{es}}$	$\mathbf{Y}_{\mathbf{es}}$	$\mathbf{Y}_{\mathbf{es}}$	\mathbf{Yes}	\mathbf{Yes}	\mathbf{Yes}
Occupation FE	$\mathbf{Y}_{\mathbf{es}}$	\mathbf{Yes}	\mathbf{Yes}	$\mathbf{Y}_{\mathbf{es}}$	\mathbf{Yes}	$\mathbf{Y}_{\mathbf{es}}$	Yes	\mathbf{Yes}	\mathbf{Yes}
Industry FE	$\mathbf{Y}_{\mathbf{es}}$	$\mathbf{Y}_{\mathbf{es}}$	\mathbf{Yes}	$\mathbf{Y}_{\mathbf{es}}$	\mathbf{Yes}	$\mathbf{Y}_{\mathbf{es}}$	\mathbf{Yes}	$\mathbf{Y}_{\mathbf{es}}$	\mathbf{Yes}
Firm Size FE	$\mathbf{Y}_{\mathbf{es}}$	\mathbf{Yes}	$\mathbf{Y}_{\mathbf{es}}$	\mathbf{Yes}	$\mathbf{Y}_{\mathbf{es}}$	\mathbf{Yes}	$\mathbf{Y}_{\mathbf{es}}$	\mathbf{Yes}	$\mathbf{Y}_{\mathbf{es}}$